UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 OR 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): February 16, 2024

INPIXON

(Exact name of registrant as specified in its charter)

Nevada	001-36404	88-0434915		
(State or other jurisdiction	(Commission File Number)	(I.R.S. Employer		
of incorporation)		Identification No.)		
2479 E. Bayshore Road, Suite 195				
Palo Alto, CA		94303		
(Address of principal executive offices)	(Zip Code)		
Registra	ant's telephone number, including area code: (408) 70	02-2167		
	N/A			
(Form	mer name or former address, if changed since last rep	port)		
Check the appropriate box below if the Form 8-K is intended to	o simultaneously satisfy the filing obligation of the I	Registrant under any of the following provisions:		
Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)				
□ Soliciting material pursuant to Rule 14a-12 under the Exc	hange Act (17 CFR 240.14a-12)			
□ Pre-commencement communications pursuant to Rule 14d	d-2(b) under the Exchange Act (17 CFR 240.14d-2(b	p))		
□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))				
Securities registered pursuant to Section 12(b) of the Act:				
Title of Each Class	Trading Symbol(s)	Name of Each Exchange on Which Registered		
Common Stock	INPX	The Nasdaq Capital Market		
Indicate by check mark whether the registrant is an emerging the Securities Exchange Act of 1934 (§240.12b-2 of this chapt		rities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of		
Emerging growth company □				
If an emerging growth company, indicate by check mark if th accounting standards provided pursuant to Section 13(a) of the		sition period for complying with any new or revised financial		

Item 1.01 Entry into a Material Definitive Agreement.

As previously reported in a current report on Form 8-K filed with the Securities and Exchange Commission (the "SEC") on July 25, 2023, Inpixon entered into an Agreement and Plan of Merger, dated July 24, 2023 (as amended from time to time, the "Merger Agreement"), by and among Inpixon, Superfly Merger Sub Inc., a wholly-owned subsidiary of Inpixon, and XTI Aircraft Company ("XTI"). As a condition to closing the merger transaction contemplated by the Merger Agreement, Inpixon is required to complete the divestiture of any business lines and other assets and liabilities that are not associated with its real time location services and analytics business, including its Shoom, SAVES and Game Your Game lines of business and investment securities, as applicable by any lawful means, including a sale to one or more third parties, spin off, plan of arrangement, merger, reorganization, or any combination of the foregoing (the "Solutions Divestiture").

As part of the Solutions Divestiture, on February 16, 2024, Inpixon entered into an Equity Purchase Agreement (the "Equity Purchase Agreement") to divest the businesses held by Grafiti LLC, a wholly-owned subsidiary of Inpixon (the "Covered Business"), by transferring 100% of the equity interest in Grafiti LLC to Grafiti Group LLC ("Purchaser"), a holding company controlled by Inpixon's director and Chief Executive Officer, Nadir Ali. The Covered Business includes assets and liabilities primarily relating to Inpixon's Saves, Shoom and Game Your Game business, including 100% of the equity interests of Inpixon India, Grafiti GmbH (previously Inpixon Gmbh) and Game Your Game, Inc., and excludes Inpixon Limited.

Pursuant to the Equity Purchase Agreement, Purchase from Inpixon 100% of the equity interest in Grafiti LLC for a minimum purchase price of \$1,000,000 paid in two annual cash installments of \$500,000 due within 60 days after December 31, 2024 and 2025. The purchase price and annual cash installment payments will be (i) increased for 50% of net income after taxes, if any, from the operations of Grafiti LLC for the years ended December 31, 2024 and 2025; (ii) decreased for the amount of transaction expenses assumed, if any; and (iii) increased or decreased by the amount the working capital of Grafiti LLC on the closing balance sheet is greater or less than \$1,000,000.

The Equity Purchase Agreement includes customary representations and warranties of Inpixon, Grafiti LLC and the Purchaser, as well as customary covenants and additional agreements, including a transition services agreement which will govern the parties' respective rights and obligations with respect to the provision of certain transition services following the consummation of the transactions pursuant to the Equity Purchase Agreement (the "Closing"). Additionally, all representations, warranties and pre-Closing covenants of Inpixon and Grafiti LLC will not survive the Closing.

The foregoing description of the Equity Purchase Agreement and the contemplated transactions therein does not purport to be complete and is qualified in its entirety by the terms and conditions of the Equity Purchase Agreement, a copy of which is attached hereto as Exhibit 2.1 and is incorporated herein by reference.

The Equity Purchase Agreement contains representations, warranties and covenants that the parties to the Equity Purchase Agreement made to each other as of the date of the Equity Purchase Agreement or other specific dates. The assertions embodied in those representations, warranties and covenants were made for purposes of the contract among the parties and are subject to important qualifications and limitations agreed to by the parties in connection with negotiating the Equity Purchase Agreement. The Equity Purchase Agreement has been attached to provide investors with information regarding its terms and is not intended to provide any other factual information about Inpixon, Grafiti LLC and the Purchaser to the Equity Purchase Agreement. In particular, the representations, warranties, covenants and agreements contained in the Equity Purchase Agreement, which were made only for purposes of the Equity Purchase Agreement and as of specific dates, were solely for the benefit of the parties to the Equity Purchase Agreement, may be subject to limitations agreed upon by the contracting parties (including being qualified by confidential disclosures made for the purposes of allocating contractual risk between the parties to the Equity Purchase Agreement instead of establishing these matters as facts) and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors and reports and documents filed with the SEC. Investors should not rely on the representations, warranties, covenants and agreements and other terms of the Equity Purchase Agreement may be subject to subsequent waiver or modification. Moreover, information concerning the subject matter of the representations and warranties and other terms may change after the date of the Equity Purchase Agreement, which subsequent information may or may not be fully reflected in Inpixon's public disclosures.

Item 2.01 Completion of Acquisition or Disposition of Assets.

On February 21, 2024, Inpixon completed the disposition of the Covered Business pursuant to the Equity Purchase Agreement. To the extent required by Item 2.01, the disclosure set forth in Item 1.01 above is incorporated by reference in this Item 2.01.

Item 8.01 Other Events.

In connection with the transactions contemplated in the Merger Agreement, Inpixon previously filed (i) XTI's unaudited financial statements as of September 30, 2023 and for the three and nine months ended September 30, 2023 and 2022, including the accompanying notes thereto, and (ii) the unaudited pro forma condensed combined balance sheet of Inpixon and XTI as of September 30, 2023 and the unaudited pro forma condensed combined statement of operations of Inpixon and XTI for the nine months ended September 30, 2023 and for the year ended December 31, 2022 (the "September 30, 2023 Pro Forma Financial Information"), as Exhibit 99.1 and Exhibit 99.2, respectively, in a Current Report on Form 8-K filed with the Securities and Exchange Commission (the "SEC") on December 15, 2023 (the "December 2023 Form 8-K").

This Current Report on Form 8-K is being filed to update the September 30, 2023 Pro Forma Financial Information of Inpixon and XTI (as updated, the "Updated September 30, 2023 Pro Forma Financial Information") due to certain adjustments made subsequent to the filing of the December 2023 Form 8-K in connection with recent developments, including, but not limited to, Inpixon's anticipated reverse stock split, the anticipated conversion of outstanding debt and certain other liabilities into equity securities by XTI and Inpixon, and the consummation of a proposed financing of equity securities by XTI. To the extent that information in the Updated September 30, 2023 Pro Forma Financial Information contained in Exhibit 99.1 hereto differs from or updates information contained in the September 30, 2023 Pro Forma Financial Information contained in Exhibit 99.2 filed with the December 2023 Form 8-K, the information contained in Exhibit 99.1 hereto shall supersede or supplement the information in Exhibit 99.2 filed with the December 2023 Form 8-K.

The Updated Pro Forma Financial Information included in this Current Report on Form 8-K is being filed for purposes of incorporating such information by reference into one or more registration statements filed or to be filed by Inpixon.

Important Information About the Proposed Transaction and Where to Find It

This Current Report on Form 8-K and the exhibits attached hereto and information incorporated herein relate to a proposed transaction between XTI and Inpixon pursuant to an agreement and plan of merger, dated as of July 24, 2023 (as amended from time to time), by and among Inpixon, Superfly Merger Sub Inc. and XTI (the "proposed transaction"). Inpixon filed a registration statement on Form S-4 with the U.S. Securities and Exchange Commission ("SEC") which was declared effective on November 13, 2023 in connection with the proposed transaction. A proxy statement/prospectus was sent to all Inpixon stockholders as of October 24, 2023, the record date established for voting on the transaction, and to the stockholders of XTI.

Investors and security holders are urged to read the registration statement, the proxy statement/prospectus, and all other relevant documents filed or that will be filed with the SEC in connection with the proposed transaction because they contain important information about Inpixon, XTI and the proposed transaction. Investors and securityholders will be able to obtain free copies of the registration statement, the proxy statement/prospectus and all other relevant documents filed or that will be filed with the SEC by Inpixon through the website maintained by the SEC at www.sec.gov.

The documents filed by Inpixon with the SEC also may be obtained free of charge at Inpixon's website at www.inpixon.com or upon written request to: Inpixon, 2479 E. Bayshore Road, Suite 195, Palo Alto, CA 94303.

NEITHER THE SEC NOR ANY STATE SECURITIES REGULATORY AGENCY HAS APPROVED OR DISAPPROVED THE TRANSACTIONS DESCRIBED IN THIS COMMUNICATION, PASSED UPON THE MERITS OR FAIRNESS OF THE TRANSACTION OR RELATED TRANSACTIONS OR PASSED UPON THE ADEQUACY OR ACCURACY OF THE DISCLOSURE IN THIS COMMUNICATION. ANY REPRESENTATION TO THE CONTRARY CONSTITUTES A CRIMINAL OFFENSE.

Forward-Looking Statements

This Current Report on Form 8-K and the exhibits attached hereto and information incorporated herein contain certain "forward-looking statements" within the meaning of the United States Private Securities Litigation Reform Act of 1995, Section 27A of the Securities Act of 1933, as amended (the "Securities Act"), and Section 21E of the Securities Exchange Act of 1934, as amended. All statements other than statements of historical fact contained in this Current Report on Form 8-K, including statements regarding the Inpixon's anticipated reverse stock split, the anticipated conversion of outstanding debt and certain other liabilities into equity securities by XTI and Inpixon, and the consummation of a proposed financing of equity securities by XTI, are forward-looking statements.

Some of these forward-looking statements can be identified by the use of forward-looking words, including "may," "should," "expect," "intend," "will," "estimate," "anticipate," "believe," "predict," "plan," "targets," "projects," "could," "would," "continue," "forecast" or the negatives of these terms or variations of them or similar expressions. All forward-looking statements are subject to risks, uncertainties, and other factors which could cause actual results to differ materially from those expressed or implied by such forward-looking statements. All forward-looking statements are based upon estimates, forecasts and assumptions that, while considered reasonable by Inpixon and its management, and XTI and its management, as the case may be, are inherently uncertain and many factors may cause the actual results to differ materially from current expectations which include, but are not limited to:

- the risk that the proposed transaction may not be completed in a timely manner or at all, which may adversely affect the price of Inpixon's securities;
- the failure to satisfy the conditions to the consummation of the proposed transaction;
- the occurrence of any event, change or other circumstance that could give rise to the termination of the merger agreement;
- the adjustments permitted under the merger agreement to the exchange ratio that could result in XTI shareholders or Inpixon shareholders owning less of the post-combination company than expected;
- the effect of the announcement or pendency of the proposed transaction on Inpixon's and XTI's business relationships, performance, and business generally;
- the risks that the proposed transaction disrupts current plans of Inpixon and XTI and potential difficulties in Inpixon's and XTI's employee retention as a result of the proposed transaction;
- the outcome of any legal proceedings against XTI or against Inpixon related to the merger agreement or the proposed transaction;
- failure to realize the anticipated benefits of the proposed transaction;
- the inability to meet and maintain the listing of Inpixon's securities (or the securities of the post-combination company) on Nasdaq;
- the risk that the price of Inpixon's securities (or the securities of the post-combination company) may be volatile due to a variety of factors, including changes in the highly competitive industries in which Inpixon and XTI operate,

- the inability to implement business plans, forecasts, and other expectations after the completion of the proposed transaction, and identify and realize additional opportunities;
- variations in performance across competitors, changes in laws, regulations, technologies that may impose additional costs and compliance burdens on Inpixon and XTI's operations, global supply chain disruptions and shortages,
- national security tensions, and macro-economic and social environments affecting Inpixon and XTI's business and changes in the combined capital structure;
- the risk that XTI has a limited operating history, has not yet manufactured any non-prototype aircraft or delivered any aircraft to a customer, and XTI and its current
 and future collaborators may be unable to successfully develop and market XTI's aircraft or solutions, or may experience significant delays in doing so;
- the risk that XTI is subject to the uncertainties associated with the regulatory approvals of its aircraft including the certification by the Federal Aviation Administration, which is a lengthy and costly process;
- the risk that the post-combination company may never achieve or sustain profitability;
- the risk that XTI, Inpixon and the post-combination company may be unable to raise additional capital on acceptable terms to finance its operations and remain a going concern;
- the risk that the post-combination company experiences difficulties in managing its growth and expanding operations;
- the risk that XTI's conditional pre-orders (which include conditional aircraft purchase agreements, non-binding reservations, and options) are canceled, modified, delayed or not placed and that XTI must return the refundable deposits;
- the risks relating to long development and sales cycles, XTI's ability to satisfy the conditions and deliver on the orders and reservations, its ability to maintain quality control of its aircraft, and XTI's dependence on third parties for supplying components and potentially manufacturing the aircraft;
- the risk that other aircraft manufacturers develop competitive VTOL aircraft or other competitive aircraft that adversely affect XTI's market position;
- the risk that XTI's future patent applications may not be approved or may take longer than expected, and XTI may incur substantial costs in enforcing and protecting its intellectual property;
- the risk that XTI's estimates of market demand may be inaccurate;
- the risk that XTI's ability to sell its aircraft may be limited by circumstances beyond its control, such as a shortage of pilots and mechanics who meet the training standards, high maintenance frequencies and costs for the sold aircraft, and any accidents or incidents involving VTOL aircraft that may harm customer confidence; and
- other risks and uncertainties set forth in the sections entitled "Risk Factors" and "Cautionary Note Regarding Forward-Looking Statements" in Inpixon's Annual Report on Form 10-K for the year ended December, 31, 2022, which was filed with the SEC on April 17, 2023 (the "2022 Form 10-K"), the Quarterly Reports on Form 10-Q for the quarterly periods filed thereafter and in the section entitled "Risk Factors" in XTI's periodic reports filed pursuant to Regulation A of the Securities Act including XTI's Annual Report on Form 1-K for the year ended December 31, 2022, which was filed with the SEC on July 13, 2023 (the "2022 Form 1-K"), as such factors may be updated from time to time in Inpixon's and XTI's filings with the SEC, the registration statement on Form S-4 and the proxy statement/prospectus contained therein. These filings identify and address other important risks and uncertainties that could cause actual events and results to differ materially from those contained in the forward-looking statements.

Nothing in this Current Report on Form 8-K should be regarded as a representation by any person that the forward-looking statements set forth herein will be achieved or that any of the contemplated results of such forward-looking statements will be achieved. You should not place undue reliance on forward-looking statements, which speak only as of the date they are made. Neither Inpixon nor XTI gives any assurance that either Inpixon or XTI or the post-combination company will achieve its expected results. Neither Inpixon nor XTI undertakes any duty to update these forward-looking statements, except as otherwise required by law.

No Offer or Solicitation

This Current Report on Form 8-K is not a proxy statement or solicitation of a proxy, consent or authorization with respect to any securities or in respect of the proposed transaction and is not intended to and does not constitute an offer to sell or the solicitation of an offer to buy, sell or solicit any securities or any proxy, vote or approval, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be deemed to be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act.

Item 9.01 Financial Statements and Exhibits.

(b) Pro forma financial information

The pro forma financial statements reflecting the disposition of the Covered Business pursuant to the Equity Purchase Agreement, to the extent required by this item, will be filed by amendment to this Current Report on Form 8-K.

Additionally, the unaudited pro forma condensed combined balance sheet of Inpixon and XTI as of September 30, 2023 and the unaudited pro forma condensed combined statement of operations of Inpixon and XTI for the nine months ended September 30, 2023 and for the year ended December 31, 2022 are attached herewith as Exhibit 99.1 to this Current Report on Form 8-K and incorporated herein by reference. These unaudited pro forma condensed combined financial statements are for informational purposes only. They do not purport to indicate the results that would have been obtained had the merger between Inpixon and XTI actually been completed on the assumed date or for the periods presented, or which may be realized in the future. The pro forma adjustments are based on the information currently available and the assumptions and estimates underlying the pro forma adjustments are described in the notes accompanying the pro forma financial information. Actual results may differ materially from the assumptions within the accompanying unaudited pro forma condensed combined financial information.

(d) Exhibits.

Exhibit No.	Description
2.1*	Equity Purchase Agreement, dated as of February 16, 2024, by and among Inpixon, Grafiti LLC and Grafiti Group LLC.
99.1	Unaudited pro forma condensed combined financial statements of Inpixon and XTI Aircraft Company.
104	Cover Page Interactive Data File (embedded within the Inline XBRL document).

^{*} The exhibits and schedules to the Equity Purchase Agreement have been omitted from this filing pursuant to Item 601(b)(2) of Regulation S-K. Inpixon agrees to furnish copies of any of such exhibits or schedules to the SEC upon request; provided, however, that Inpixon may request confidential treatment pursuant to Rule 24b-2 of the Securities Exchange Act of 1934, as amended, for any schedules so furnished.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

INPIXON

Date: February 23, 2024 By: /s/ Nadir Ali

Name: Nadir Ali
Title: Chief Executive Officer

EQUITY PURCHASE AGREEMENT

DATED AS OF FEBRUARY 16, 2024

BY AND AMONG

INPIXON,

GRAFITI LLC,

AND

GRAFITI GROUP LLC

TABLE OF CONTENTS

		Page
ARTICLE I	Definitions	
1.1	Definitions	1
ARTICLE I	I Purchase and Sale of Purchased Interest; Closing and Manner of Payment	7
2.1	Agreement to Purchase and Sell Purchased Interest	7
2.2	Purchase Price	7
2.3	Working Capital Adjustment	8
2.4	Determination of Working Capital and Certain Other Items	8
2.5	Disputes Regarding Closing Balance Sheet	8
2.6	Intentionally Omitted.	9
2.7	Intentionally Omitted.	9
2.8	Time and Place of Closing	9
2.9	Purchase Price Installment Payments	9
2.10	Withholding	10
ARTICLE I	II Representations and Warranties	
		10
3.1	General Statement	10
3.2	Representations and Warranties of Buyer	10
3.3	Representations and Warranties of the Company	13
3.4	Representations and Warranties of Seller	16
3.5	Limitation on Warranties	17
3.6	Definition of Knowledge	17
ARTICLE I	V Conduct Prior to the Closing	17
4.1	General	17
4.2	Company's and Seller's Obligations	17
4.3	Buyer's Obligations	19
4.4	Joint Obligations	19
ARTICLE V	Conditions to Closing	20
		20
5.1	Conditions to the Company's and Seller's Obligations	20
5.2	Conditions to Buyer's Obligations	20
5.3	Frustration of Closing Conditions	21
5.4	Waiver of Conditions	21
A DELCT E X		21
ARTICLE V	T Closing	21
(.1	France of December	21
6.1	Form of Documents	21
6.2 6.3	Buyer's Deliveries Seller's Deliveries	21 21
0.3	School & Dehveries	21

ARTICLE VII Post-Closing Agreements		22
7.1	Post-Closing Agreements	22
7.2	Inspection of Records	22
7.3	Intentionally Omitted	22
7.4	Tax Matters	22
7.5	Employee Matters	24
7.6	Further Assurances	24
A DELCT E A	ATHLO ' I D. I'	25
ARTICLE V	VIII Survival; Remedies	25
8.1	Survival	25
8.2	Remedies	25
0.2	Remedies	23
ARTICLET	IX Effect of Termination/Proceeding	25
THE TODA		25
9.1	General	25
9.2	Right to Terminate	26
9.3	Certain Effects of Termination	26
9.4	Remedies	27
9.5	Right to Damages	27
	· ·	
ARTICLE X	X Miscellaneous	27
10.1	Publicity	27
10.2	Notices	27
10.3		28
10.4	Entire Agreement	29
10.5	Projections	29
10.6	Non-Waiver	29
10.7		29
10.8		30
10.9		30
10.10		30
10.11		30
10.12		30
10.13		31
10.14	4 Consent to Jurisdiction	31
10.15		31
10.16		31
10.17	7 Release and Indemnity	32

EQUITY PURCHASE AGREEMENT

THIS EQUITY PURCHASE AGREEMENT (this "Agreement") is made as of February 16, 2024, by and among Grafiti LLC, a Nevada limited liability company (the "Company"), Inpixon, a Nevada corporation ("Seller"), and Grafiti Group LLC, a Nevada limited liability company ("Buyer").

RECITALS

- A. Seller is engaged in the business of (i) operating an advertising management platform consisting of digital solutions (eTearsheets; eInvoice, adDelivery) and cloud-based applications and analytics for the advertising, media and publishing industries, collectively referred to publicly by Seller and the Company as Shoom, (ii) providing comprehensive sets of data analytics and statistical visualization solutions for engineers and scientists referred to publicly by Seller and the Company as SAVES, excluding such solutions provided by Inpixon Ltd., a United Kingdom limited company, and (iii) managing the Company's set of applications, technology and data analytics related to the Company's provision of sports performance enhancing solutions referred to publicly by Seller and the Company as Game Your Game (GYG) (the "Business").
- B. Seller and Company have entered into that certain Contribution, Assignment, and Assumption Agreement dated December 21, 2023 (the "Contribution Agreement") pursuant to which Seller has contributed and assigned to the Company and the Company has accepted the Transferred Assets and the Transferred Liabilities (as such terms are defined in the Contribution Agreement) associated with the Business.
 - C. Seller owns all of the issued and outstanding limited liability company interests of the Company (collectively, the "Interest").
- D. Buyer desires to purchase from Seller, and Seller desires to sell to Buyer, all of the Interest (the "Purchased Interest"), on the terms and subject to the conditions herein contained.

AGREEMENTS

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

ARTICLE I Definitions

- 1.1 <u>Definitions</u>. For purposes of this Agreement, the following terms have the meanings set forth below.
- "Affiliate" with respect to any Person means any other Person who directly or indirectly Controls, is Controlled by, or is under common Control with such Person including, in the case of any Person who is an individual, his or her spouse, any of his or her descendants (lineal or adopted) or ancestors, and any of their spouses.

- "Agreement" has the meaning set forth in the introductory paragraph.
- "Arbitrating Accountant" means the independent accounting or consulting firm mutually acceptable to Seller and Buyer or, if they are unable to agree, Seller and Buyer's respective accountants acting jointly.
 - "Business" has the meaning set forth in the recitals to this Agreement.
- "Business Day" means any day other than Saturday, Sunday and any day that is a legal holiday or a day on which banking institutions in New York, New York are permitted or required to be closed.
 - "Buyer" has the meaning set forth in the introductory paragraph.
 - "Buyer Closing Certificate" has the meaning set forth in Section 5.1(c).
 - "Buyer Indemnitor" means Buyer and its successors and assigns, and the term "Buyer Indemnitor" means any one of the foregoing Buyer Indemnitors.
 - "Buyer Prepared Tax Return" has the meaning set forth in Section 7.4(a).
 - "Closing" has the meaning set forth in Section 2.8.
 - "Closing Balance Sheef" means the unaudited balance sheet of the Company as of 11:59 p.m. (Pacific time) on the day immediately preceding the Closing Date.
 - "Closing Date" has the meaning set forth in Section 2.8.
 - "Code" means the Internal Revenue Code of 1986, as amended.
 - "Company" has the meaning set forth in the recitals to this Agreement.
 - "Company Closing Certificate" has the meaning set forth in Section 5.2(c).
 - "Company IP" has the meaning set forth in Section 3.3(q)(ii).
 - "Contribution Agreement" has the meaning set forth in the recitals to this Agreement.
- "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through ownership of securities, by contract or otherwise, and the term "Controlled" has the meaning correlative thereto.
 - "Delivery Date" means the date on which the Closing Balance Sheet has been delivered.
- "Disclosure Schedule" means the schedules delivered by Seller to Buyer concurrently herewith and identified by the parties as the Disclosure Schedule (as amended, modified or supplemented in accordance with the terms hereof).

- "Dispute" means any dispute regarding the items or amounts reflected on the Closing Balance Sheet and affecting the calculation of the Purchase Price.
- "Dispute Notice" means a written notice of a Dispute presented to Seller within the Dispute Period.
- "Dispute Period" means the period beginning on the Delivery Date and ending at 5:00 p.m. (Pacific time) on the date thirty (30) days after the Delivery Date.
- "Disputed Items" has the meaning set forth in Section 2.5(a).
- "Family Members" means a Person's spouse, parents, children, siblings, mothers and fathers-in-law, sons and daughters-in-law, brothers and sisters-in-law, and anyone (other than domestic employees) who share such Person's home.
 - "First Installment" has the meaning set forth in Section 2.9(a).
 - "First Installment Period" means the period from January 1, 2024 through December 31, 2024.
- "GAAP" means United States generally accepted accounting principles applied in a manner consistent with the accounting principles and practices applied in the preparation of Seller's financial statements prior to the Closing as filed with the U.S. Securities and Exchange Commission.
- "Governmental Authority" means the United States or any state, provincial, local or foreign government, or any subdivision, agency or authority of any thereof having competent jurisdiction over any of the Company, Buyer, Seller or the transactions contemplated by this Agreement, as applicable.
- "Indebtedness" means, without duplication, to the extent not included in Working Capital, the sum of the following items of the Company as of the Closing: (a) all indebtedness for borrowed money (including the principal amount thereof and the amount of accrued and unpaid interest thereon) of the Company, whether or not represented by bonds, debentures, notes or other securities, for the repayment of money borrowed, whether owing to banks, financial institutions or otherwise (but excluding trade payables, accrued expenses and the like); (b) all guaranties of the Company in respect of indebtedness for borrowed money of Persons other than the Company; and (c) all premiums, fees, penalties, change of control payments or other amounts due in respect of any of the foregoing as a result of the consummation of the transactions contemplated by this Agreement. For purposes of this Agreement (and notwithstanding anything expressed or implied herein to the contrary), Transaction Expenses, leases (whether operating lease or capitalized leases) and undrawn amounts (or contingent reimbursement obligations) under any outstanding letters of credit shall not be deemed to be Indebtedness.

"Indemnified Employees" means all present, and former directors, managers, officers, employees and agents of the Company and Seller (to the extent such employees and agents of Seller have provided services in connection with the Business), and the term "Indemnified Employee" means any one of the foregoing Indemnified Employees.

"Installment Measurement Date" with respect to any Installment Period means the last day ending such Installment Period.

"Installments" has the meaning set forth in Section 2.9(a).

"Installment Periods" means the First Installment Period and the Second Installment Period, as applicable.

"Intellectual Property" means (a) foreign and domestic patents and patent applications; (b) Internet domain names, trademarks, service marks, trade dress, trade names, logos and corporate or company names (both foreign and domestic) and registrations and applications for registration thereof together with all of the goodwill associated therewith; (c) copyrights (registered or unregistered) and copyrightable works (both foreign and domestic) and registrations and applications for registration thereof; (d) computer software, data, databases and documentation thereof, including rights to third party software used in the Business; and (e) trade secrets and other confidential information (including ideas, formulas, compositions, inventions, know how, manufacturing and production processes and techniques, research and development information, drawings, specifications, designs, plans, proposals, technical data, copyrightable works, financial and marketing plans and customer and supplier lists and information).

"Interest" has the meaning set forth in the recitals to this Agreement.

"IRS" means the Internal Revenue Service.

"Liens" means all options, proxies, voting trusts, voting agreements, judgments, pledges, charges, escrows, rights of first refusal or first offer, mortgages, indentures, claims, transfer restrictions, liens, equities, security interests and other encumbrances of every kind and nature whatsoever, whether arising by agreement, operation of law or otherwise, other than (a) restrictions on the offer and sale of securities under federal and state securities laws and (b) any Permitted Liens.

"Material Adverse Effect" means a material adverse effect on the long-term business or financial condition of the Company, taken as a whole, provided that the foregoing shall not include any event, circumstance, change, occurrence, fact or effect resulting from or relating to (a) changes in economic conditions generally or in any region in which the Company or the Business operate; (b) changes in United States or global financial markets in general; (c) changes, occurrences or developments in or related to the general industry or industries (or portions thereof) in which the Company or the Business operate or are materially related thereto; (d) changes in law, accounting standards or any authoritative interpretations thereof; (e) any action taken or failed to be taken by the Company or Seller or any of their Affiliates or representatives at the request of Buyer or that is required or contemplated by this Agreement; (f) a failure to meet the Company's projections, or any changes in the prices or availability of labor, and/or any other items or services used in the Business; (g) the identity of, or any action taken by, Buyer or any of its Affiliates or representatives; (h) the announcement of, publicity related to, and/or performance of this Agreement and the other transactions contemplated by this Agreement, including termination of, reduction in or similar negative impact on relationships, contractual or otherwise, with any customers, suppliers, representatives, partners, officers or employees of the Business; (i) any actions required under this Agreement to obtain any approval or authorization required under applicable laws for the consummation of the transactions contemplated by this Agreement; (j) acts of war (whether or not declared), armed hostilities, sabotage or terrorism occurring after the date of this Agreement or the continuation, escalation or worsening of any such acts of war, armed hostilities, sabotage or terrorism threatened or underway as of the date of this Agreement; (k) earthquakes, hurrican

"Material Contracts" means all of the contracts, leases, agreements and instruments listed in Section 3.3(1) of the Disclosure Schedule.

"Net Income After Taxes" means with respect to an Installment Period, the net income of the Company for such period, calculated in accordance with GAAP, after income tax expense, plus income tax credits/receivables, provided that any amounts paid in cash as salary, guaranteed payments, or fees to Nadir Ali or his Family Members or any of their respective Affiliates during the applicable Installment Period shall not be considered an expense for purposes of calculating net income of the Company for such Installment Period, provided that any such payments made to Affiliates of Mr. Ali on arms-length fair market terms shall be considered an expense for purposes of calculating net income hereunder.

"Net Income After Taxes Determination" has the meaning set forth in Section 2.9(a).

"NRF" has the meaning set forth in Section 10.16.

"Permits" means all licenses, permits, registrations and government approvals other than the Environmental Permits.

"Permitted Liens" means: (a) statutory Liens for current Taxes, assessments and other charges by Governmental Authorities that are not yet due and payable (or that may be paid without interest or penalties) or that are being contested in good faith; (b) statutory liens of landlords, carriers, warehousemen, mechanics and materialmen incurred in the ordinary course of business for sums not yet due; (c) Liens incurred or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance and other types of social security or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, government contracts, performance and return of money bonds and similar obligations; (d) minor irregularities of title which do not in the aggregate materially detract from the value or use of the Company's assets; (e) such covenants, conditions, restrictions, easements, encroachments and other encumbrances that would be shown by a current, accurate survey or physical inspection of any leased real estate; (f) zoning, building codes and other land use laws regulating the use or occupancy of real property or the activities conducted thereon which are imposed by any Governmental Authority having jurisdiction over real property; (g) a lessor's interest in, and any mortgage, pledge, security interest, encumbrance, Lien (statutory or other) or conditional sale agreement on or affecting a lessor's interest in, any leased real estate; or (h) Liens or encumbrances or matters caused by, or resulting from, the actions of Buyer or any of its agents, employees or Affiliates.

- "Person" means any individual, corporation, partnership, limited liability company, joint venture, association, bank, trust company, trust or other entity, whether or not a legal entity, or any Governmental Authority.
- "Personal Information" means any information which allows the identification of a natural person, including, without limitation, a natural person's name, street address, email address, telephone number, photograph, social security number or tax identification number, driver's license number, credit card number, bank information or customer or account number.
- "Pre-Closing Tax Period" means any taxable period, or portion thereof, ending on or before the Closing Date, including, for the avoidance of doubt, the portion of any Straddle Period ending on the Closing Date.
 - "Purchase Price" has the meaning set forth in Section 2.2.
 - "Purchased Interest" has the meaning set forth in the recitals to this Agreement.
 - "Released Claims" has the meaning set forth in Section 10.17.
 - "Released Person" has the meaning set forth in Section 10.17.
 - "Releasing Person" has the meaning set forth in Section 10.17.
 - "Remaining Disputed Items" has the meaning set forth in Section 2.5(b).
 - "Second Installment" has the meaning set forth in Section 2.9(a).
 - "Second Installment Period" means the period from January 1, 2025 through December 31, 2025.
 - "Securities Act" means the Securities Act of 1933, as amended.
 - "Seller" has the meaning set forth in the introductory paragraph.
- "Seller Indemnitees" means Seller and its respective Affiliates, agents, representatives, heirs and assigns, and each of their respective directors, managers, officers, members, shareholders, partners, agents, representatives, successors, heirs and assigns, and the term "Seller Indemnitee" means any one of the foregoing Seller Indemnitees.
- "Software" means any and all: (a) computer programs, including any and all software implementation of algorithms, models and methodologies whether in source code or object code; (b) databases and computations, including any and all data and collections of data; (c) documentation, including user manuals and training materials, relating to any of the foregoing; and (d) content and information contained in any web site.
 - "Straddle Period" means any taxable period that includes (but does not end on) the Closing Date.

- "Taxes" means all federal, state, foreign and local income, sales, use, ad valorem, transfer or other taxes, fees, assessments or charges of any kind, together with any interest and any penalties with respect thereto, and the term "Tax" means any one of the foregoing Taxes.
- "Tax Return" means any return, declaration, report, claim for refund or other document or form relating to Taxes filed or required to be filed with any Governmental Authority in connection with the determination, assessment or collection of any Tax, including any schedule or attachment thereto, and including any amendment thereof.
 - "Termination Date" has the meaning set forth in Section 9.2(b).
- "Transaction Agreements" means this Agreement, the Contribution Agreement, the Transition Services Agreement and each other contract, certificate, instrument, agreement or other document executed and delivered in connection herewith or the transactions contemplated hereby or thereby.
- "Transaction Expenses" means all of the Company's and Sellers' expenses incurred in connection with the preparation, execution and consummation of this Agreement, including attorneys', accountants' and other advisors' fees and expenses payable by the Company or the Seller as of the Closing or not otherwise included in Working Capital.
 - "Transfer Taxes" has the meaning set forth in Section 10.3.
- "Working Capital" means the excess of the Company which are treated as current assets (exclusive of inventory), minus the liabilities of the Company which are treated as current liabilities (exclusive of Transaction Expenses), all determined in the manner set forth in Section 2.4.
- "Working Capital Adjustment" means (a) a positive amount equal to the amount by which the Working Capital of the Company set forth in the Closing Balance Sheet is greater than \$1,000,000, or (b) a negative amount equal to the amount by which the Working Capital of the Company set forth in the Closing Balance Sheet is less than \$1,000,000.
- "XTI Merger Agreement" means the Agreement and Plan of Merger, dated as of July 24, 2023, by and among XTI Aircraft Company, Seller and Superfly Merger Sub Inc., as amended from time to time by the parties thereto.

ARTICLE II

Purchase and Sale of Purchased Interest; Closing and Manner of Payment

- 2.1 Agreement to Purchase and Sell Purchased Interest. On the terms and subject to the conditions contained in this Agreement, at the Closing, Seller shall sell, convey, assign, transfer and deliver to Buyer, and Buyer shall purchase, the Purchased Interest, free and clear of any and all Liens.
- 2.2 <u>Purchase Price</u>. The aggregate purchase price for all of the Purchased Interest (the "*Purchase Price*") shall be equal to: (a) the aggregate amount of one million (\$1 million); <u>plus (b)</u> fifty percent (50%) of Net Income After Taxes, if any, calculated as of each Installment Measurement Date; <u>minus (c)</u> the amount of Transaction Expenses assumed or otherwise discharged by Company or Buyer, if any, and required to be paid by Seller pursuant to this Agreement; <u>plus</u> or <u>minus (d)</u> the amount of the Working Capital Adjustment. The Purchase Price shall be payable in cash as Installments following the Closing, by wire transfer or delivery of otherwise immediately available funds in accordance with the procedures set forth in <u>Section 2.9</u>.

- 2.3 Working Capital Adjustment. The Purchase Price will be increased or decreased on a dollar-for-dollar basis by the amount of the Working Capital Adjustment. In the event the Working Capital Adjustment is a positive number, such amount shall be added to the Purchase Price pursuant to Section 2.2. In the event the Working Capital Adjustment is a negative number, such amount shall be subtracted from the Purchase Price pursuant to Section 2.2.
- 2.4 <u>Determination of Working Capital and Certain Other Items</u>. The amounts of Indebtedness, Transaction Expenses and Working Capital shall each be determined from the Closing Balance Sheet. The Closing Balance Sheet shall be prepared by or at the direction of Seller. The Closing Balance Sheet shall be prepared in accordance with the standards set forth on <u>Schedule 2.4</u> and otherwise in accordance with GAAP. To the extent there is a conflict between GAAP and the principles and procedures set forth on <u>Schedule 2.4</u> shall control. Seller shall use commercially reasonable efforts to cause the Delivery Date to be not more than forty-five (45) days following the Closing Date. Buyer shall make available, and without cost to Seller, the books, records and personnel of the Company which Seller reasonably requires in order to prepare and deliver the Closing Balance Sheet. Buyer and Seller shall, throughout the entire period from the date of this Agreement to the Delivery Date, meet and discuss any and all financial and business matters relating to such process and the preparation of the Closing Balance Sheet.
 - 2.5 <u>Disputes Regarding Closing Balance Sheet</u>. Disputes with respect to the Closing Balance Sheet shall be resolved as follows:
- (a) Buyer shall have the Dispute Period to bring a Dispute, but only on the basis that the amounts reflected on the Closing Balance Sheet were not presented in accordance with Section 2.4 or were inaccurate. If Buyer does not give a Dispute Notice, the Closing Balance Sheet shall be deemed to have been accepted and agreed to by Buyer in the form in which it was delivered to Buyer, and shall be final and binding upon the parties hereto. If Buyer has a Dispute, Buyer shall give Seller a Dispute Notice within the Dispute Period, setting forth in reasonable detail the items and amounts in dispute (collectively, the "Disputed Items") (it being understood that all other items and amounts not so disputed shall be deemed final). Within thirty (30) days after delivery of such Dispute Notice, the parties hereto shall attempt to resolve the Disputed Items and agree in writing upon the final content of the disputed balance sheet.
- (b) If Buyer and Seller are unable to resolve any Disputed Items within the thirty (30) day period after Seller's receipt of a Dispute Notice (such items and/or amounts remaining in dispute, collectively, the "Remaining Disputed Items"), Seller and Buyer shall jointly engage the Arbitrating Accountant as arbitrator. In connection with the resolution of the Remaining Disputed Items, the Arbitrating Accountant shall have access to all documents, records, work papers, facilities and personnel necessary to perform its function as arbitrator. The Arbitrating Accountant's function shall be to resolve the Remaining Disputed Items (and only the Remaining Disputed Items) in accordance with the requirements of Section 2.4 and Schedule 2.4 and shall be bound by the definitions of Indebtedness, Transaction Expenses, Working Capital and Working Capital Adjustment, and such review shall be based solely on presentations and submissions by Buyer and Seller (and not by independent review of the Closing Balance Sheet), and upon such resolution, conform the Closing Balance Sheet accordingly. The Arbitrating Accountant shall allow Buyer and Seller to present their respective positions regarding the Remaining Disputed Items. The Arbitrating Accountant may, at its discretion, conduct a conference concerning the Remaining Disputed Items, at which conference each party shall have the right to present additional documents, materials and other information and to have present its advisors, counsel and accountants. In connection with such process, there shall be no other hearings or any oral examinations, testimony, depositions, discovery or other similar proceedings. In resolving any Remaining Disputed Item, the Arbitrating Accountant may not assign a value to any item greater than the greatest value for such item claimed by either party or less than the smallest value for such item claimed by either party. The Arbitrating Accountant shall promptly, and in any event within sixty (60) days after the date of its appointment, render its decision on the Remaining Disputed Items in writing and finalize the Closing Balance Sheet. Such written determination shall be final and binding upon the parties hereto, and judgment may be entered on the award. Upon the resolution of all Disputes, the Closing Balance Sheet shall be revised to reflect such resolution. The Arbitrating Accountant shall determine the proportion of its fees and expenses to be paid by each of Seller and Buyer, based primarily on the degree to which the Arbitrating Accountant has accepted the positions of the respective parties.

2.6 Intentionally Omitted.

2.7 Intentionally Omitted.

2.8 <u>Time and Place of Closing</u>. The transactions contemplated by this Agreement shall be consummated (the "*Closing*") remotely as of immediately prior to the closing of the transactions contemplated by the XTI Merger Agreement, or at such other place, time or date as Buyer and Seller mutually agree. The date on which the Closing occurs in accordance with the preceding sentence is referred to in this Agreement as the "*Closing Date*".

2.9 Purchase Price Installment Payments

(a) The Purchase Price, as finally determined pursuant to Section 2.2 and this Section 2.9, shall be payable by Buyer and its Affiliates in two (2) installments as follows: (i) the first such payment representing fifty percent (50%) of the aggregate Purchase Price (the "First Installment"), within sixty (60) days after the end of the First Installment Period and (ii) the second such payment representing the remaining fifty percent (50%) of the Purchase Price (the "Second Installment" and together with the First Installments"), within sixty (60) days after the end of the Second Installment Period. In no event shall the Purchase Price or any Installment payable hereunder be reduced or otherwise adjusted, and Seller shall in no way be liable to Buyer, if Net Income After Taxes yields a negative number.

(b) Upon or as promptly as practicable following the applicable Installment Measurement Date but in any event no later than forty-five (45) days following the applicable Installment Measurement Date, the Buyer shall deliver a written copy of its determination of Net Income After Taxes for the Installment Period determined in accordance with this Section 2.9 (the "Net Income After Taxes Determination") to the Seller. The Net Income After Taxes Determination shall be final, conclusive and binding on the parties hereto unless the Seller provides a written notice to the Buyer disputing such determination no later than the fifteenth (15th) calendar day after the delivery to the Sellers of the Net Income After Taxes Determination. The parties will attempt to resolve such dispute involving the Net Income After Taxes Determination in the same manner as set forth in Section 2.5. Notwithstanding anything contained herein to the contrary and in the event of such a dispute, the Buyer is still obligated to make such portion of any payments for the applicable Installment Period to the Seller that are undisputed in the manner described in this Section 2.9.

(c) During the Installment Period, the Buyer agrees (i) to operate the Business in good faith and in a manner intended to maximize Net Income After Taxes during each Installment Period, subject to and in compliance with applicable law, and (ii) not to take any actions with the intent of reducing any payments due to the Seller pursuant to this Section 2.9.

2.10 Withholding

. Buyer shall be entitled to deduct and withhold from any payment to any Person under this Agreement such amounts as it is required to deduct and withhold with respect to the making of such payment or any other Tax withholding obligation with respect to this Agreement under the Code or any provision of applicable Tax law; provided that Buyer shall use reasonable best efforts to provide Seller with a written notice of its intention to withhold or deduct at least five (5) Business Days prior to such withholding or deduction. Both the applicable payor and the applicable payee shall use commercially reasonable efforts to minimize any such Tax withholding. To the extent that amounts are so withheld or deducted by Buyer and timely paid over to the relevant Governmental Authority, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to such Person in respect of which such deduction and withholding was made.

ARTICLE III Representations and Warranties

- 3.1 General Statement. The parties make the representations and warranties to each other which are set forth in this Article III.
- 3.2 Representations and Warranties of Buyer. Buyer represents and warrants to the Company and Seller as follows:
 - (a) Organization, Existence and Good Standing. Buyer is a limited liability company duly organized, existing and in good standing, under the laws of Nevada.
- (b) <u>Power and Authority</u>. Buyer has all requisite capacity, power and authority to enter into and perform this Agreement and the other Transaction Agreements. The execution, delivery and performance of this Agreement and the other Transaction Agreements by Buyer and the consummation by Buyer of the transactions contemplated hereby have been duly and validly authorized. No other proceeding on the part of Buyer is necessary to authorize the execution, delivery and performance of this Agreement and the other Transaction Agreements by Buyer and the consummation by Buyer of the transactions contemplated hereby and thereby.
- (c) <u>Enforceability</u>. This Agreement and each other Transaction Agreement has been duly executed and delivered by Buyer and constitutes a legal, valid and binding agreement of Buyer, enforceable against Buyer in accordance with its terms, except to the extent that enforcement may be affected by laws relating to bankruptcy, reorganization, insolvency and creditors' rights and by the availability of injunctive relief, specific performance and other equitable remedies.

- (d) <u>Consents</u>. No consent, authorization, order or approval of, or filing or registration with, any Governmental Authority is required for or in connection with the consummation by Buyer of the transactions contemplated by this Agreement or the other Transaction Agreements.
- (e) <u>Conflicts Under Constituent Documents or Laws</u>. Neither the execution and delivery of this Agreement nor the other Transaction Agreements by Buyer, nor the consummation by Buyer of the transactions contemplated hereby and thereby, will conflict with or result in a breach of any of the terms, conditions or provisions of Buyer's organizational documents, or of any statute or administrative regulation, or of any order, writ, injunction, judgment or decree of any court or Governmental Authority or of any arbitration award. Buyer has no knowledge of any facts or circumstances that, in each case as to Buyer and its Affiliates, would prohibit, prevent, delay or otherwise impede: (i) Buyer's ability to acquire and/or own the Purchased Interest; or (ii) receipt of any Permits, consents, findings of qualification or suitability (of Buyer, its Affiliates and their respective officers and directors) or the like from any Governmental Authority as is necessary in connection with the transactions contemplated by this Agreement.
- (f) <u>Conflicts Under Contracts</u>. Buyer is not a party to, or bound by, any unexpired, undischarged or unsatisfied written or oral contract, agreement, indenture, mortgage, debenture, note or other instrument under the terms of which performance by Buyer according to the terms of this Agreement or the other Transaction Agreements will require a consent, approval or notice, or will result in a default or an event of acceleration, or grounds for termination, modification or cancellation, or whereby timely performance by Buyer according to the terms of this Agreement or the other Transaction Agreements may be prohibited, prevented or delayed.
- (g) <u>Funding</u>. Buyer will have, sufficient immediately available funds when due to consummate the transactions contemplated hereby on the terms contained herein, including to pay the Purchase Price and the fees and expenses of Buyer related to the transactions contemplated hereby. There is no circumstance or condition that, in the aggregate with all other circumstances and conditions, could reasonably be expected to prevent or substantially delay the availability of such funds at the times required to be paid hereunder. Buyer acknowledges and agrees that the performance of its obligations under this Agreement is not in any way contingent upon the availability of financing to Buyer.
- (h) WARN Act. Buyer has no present plans or intention to carry out, after the Closing, any plant closing or mass layoff which would require notification under, or otherwise violate the federal Worker Adjustment and Retraining Notification Act (or any similar foreign, state or local law) at any facility of the Company.

- (i) <u>Solvency</u>. Immediately after giving effect to the transactions contemplated by this Agreement, each of Buyer and the Company shall be able to pay their respective debts as they become due. Immediately after giving effect to the transactions contemplated by this Agreement, each of Buyer and the Company shall have adequate capital to carry on their respective businesses, including the Business. No transfer of property is being made and no obligation is being incurred in connection with the transactions contemplated by this Agreement and the other Transaction Agreements with the intent to hinder, delay or defraud either present or future creditors of either Buyer or the Company.
- (j) No Knowledge of Misrepresentations or Omissions. Buyer has no knowledge (i) that the representations and warranties of the Company and/or Seller in this Agreement, as modified by the Disclosure Schedule, are not true and correct in all material respects, or (ii) that there are any material errors in or material omissions from the Disclosure Schedule.
- (k) <u>Independent Investigation</u>. Buyer has conducted an independent investigation of the Company and its business operations, assets, liabilities, results of operations, condition (including, operating, environmental and financial condition) and prospects in making its determination as to the propriety of the transactions contemplated by this Agreement, and Buyer is satisfied with the results thereof. Buyer has been permitted full and complete access to the books and records, facilities, equipment, Tax Returns, contracts, insurance policies (or summaries thereof) and other properties and assets of the Company that Buyer and its representatives have desired or requested to see or review, and that Buyer and its respective representatives have had a full opportunity to meet with the officers and employees of the Company to discuss the business of the Company and have otherwise been furnished with or given full and complete access to such information about the Company and their businesses and operations as they have requested. In entering into this Agreement, Buyer has relied solely on the results of its own investigation and on the representations and warranties of the Company and Seller expressly contained in <u>Sections 3.3</u> and <u>3.4</u> of this Agreement.
- (1) <u>Investment</u>. Buyer is acquiring the Purchased Interest for its own account for investment and with no present intention of distributing or reselling such Purchased Interest or any part thereof in any transaction which would constitute a "distribution" within the meaning of the Securities Act. Buyer understands that the Purchased Interest has not been registered under the Securities Act or any state securities laws and is being transferred to Buyer, in part, in reliance on the foregoing representation.
- (m) <u>Brokers</u>. Neither Buyer nor any of its Affiliates has dealt with any Person who is entitled to a broker's commission, finder's fee, investment banker's fee or similar payment from Buyer or any of its Affiliates, or Seller or the Company, for arranging the transactions contemplated hereby or introducing the parties to each other.

- 3.3 Representations and Warranties of the Company. The Company makes the following representations and warranties to Buyer set forth in this Section 3.3 as of the date hereof. All representations and warranties of the Company are made subject to the exceptions noted in the Disclosure Schedule (whether or not a particular representation or warranty is specifically modified by the phrase "except as set forth in the Disclosure Schedule" or words of similar import). Any disclosure set forth on any particular schedule of the Disclosure Schedule shall be treated as disclosed with respect to all other schedules of the Disclosure Schedule and all other sections of this Agreement to the extent that the applicability of such item to such other schedules and such other sections of this Agreement is reasonably apparent. The inclusion of any item or fact in the Disclosure Schedule shall not be deemed an admission that such item or fact is material for the purposes of this Agreement or that such item or fact did not arise in the ordinary course of the Company's business.
- (a) Organization, Existence and Good Standing. The Company is a limited liability company duly organized, existing and in good standing under the laws of the State of Nevada.
- (b) <u>Foreign Good Standing</u>. The Company has qualified as a foreign limited liability company, and is in good standing, under the laws of all jurisdictions where the nature of its business or the nature or location of its assets requires such qualification and where the failure to so qualify would have a Material Adverse Effect.
- (c) <u>Power and Authority</u>. The Company has all necessary limited liability power and authority to carry on its business as such business is now being conducted. The Company has full limited liability company power and authority to enter into and perform this Agreement and the other Transaction Agreements. The execution, delivery and performance of this Agreement and the other Transaction Agreements by the Company and the consummation by the Company of the transactions contemplated hereby and thereby have been duly and validly approved by the sole managing member of the Company. No other limited liability company proceedings are necessary on the part of the Company to authorize (i) the execution, delivery and performance of this Agreement and the other Transaction Agreements by the Company, and (ii) the consummation by the Company of the transactions contemplated hereby and thereby.
- (d) Enforceability. Each of this Agreement has been, and the other Transaction Agreements to which the Company is or will, at the Closing, be a party, will be, duly executed and delivered by the Company and constitute a legal, valid and binding agreement of the Company, enforceable against the Company in accordance with its terms, except to the extent that enforcement may be affected by laws relating to bankruptcy, reorganization, insolvency and creditors' rights and by the availability of injunctive relief, specific performance and other equitable remedies.
- (e) <u>Consents</u>. The Company is not required to obtain or make, as applicable, any consent, authorization, order or approval of, or filing or registration with, any Governmental Authority in connection with the consummation of the transactions contemplated hereby.
- (f) <u>Conflicts Under Constituent Documents or Laws</u>. Neither the execution and delivery of this Agreement by Company, nor the consummation by Company of the transactions contemplated hereby, will conflict with or result in a breach of any of the terms, conditions or provisions of the Company's certificate of formation or limited liability company agreement, or any statute or administrative rule or regulation, or of any order, writ, injunction, judgment or decree of any court or Governmental Authority or of any arbitration award to which the Company is a party or by which the Company or any of its properties or assets is bound.

- (g) <u>Conflicts Under Contracts</u>. Except as set forth in <u>Section 3.3(g)</u> of the Disclosure Schedule, the Company is not a party to, or bound by, any unexpired, undischarged or unsatisfied Material Contract under the terms of which performance by the Company according to the terms of this Agreement will require a consent, approval or notice, or will result in a default or an event of acceleration, or grounds for termination, modification or cancellation of any right or obligation, or would prohibit, prevent or delay timely performance by the Company of this Agreement.
- (h) <u>Subsidiaries</u>. Except as set forth on Section 3.3(h), the Company does not hold or beneficially own any direct or indirect interest (whether it be common or preferred stock or any comparable ownership interest in any Person that is not a corporation), or any subscriptions, options, warrants, rights, calls, convertible securities or other agreements or commitments for any interest in any Person.
- (i) <u>Constituent Documents</u>. True and complete copies of the certificate of formation and all amendments thereto, the limited liability company agreement as amended and currently in force and all organizational records of the Company have been made available for inspection by Buyer. The organizational records of the Company contain true and complete copies of all resolutions adopted by the Company's member and/or the board of directors of the Company.
- (j) <u>Capitalization</u>. The Interest constitutes all of the issued and outstanding limited liability company interest of the Company. The Interest has been validly issued, is fully paid and nonassessable, and is owned beneficially and of record by Seller, free and clear of any and all Liens. Except as set forth on Section 3.3(j) of the Disclosure Schedule, there are no outstanding subscriptions, options, warrants, rights (including preemptive rights), calls, convertible securities or other agreements or commitments of any character relating to the issued limited liability company interests or other securities of the Company or otherwise obligating the Company to issue any securities of any kind, and there are no voting agreements with respect to any of the foregoing. The Company is under no obligation to repurchase, redeem or otherwise acquire any of its limited liability company interests or other securities.
- (k) <u>Assets</u>. The Company has good and marketable title to or, in the case of leased property, has valid leasehold interests in, all tangible personal property (including all fixtures, leasehold improvements, equipment, office, operating and other supplies and furniture) material to its business as presently conducted, free and clear of any Liens, except for Permitted Liens. The machinery, equipment and other tangible assets of the Company are in adequate condition and repair in all material respects, ordinary wear and tear excepted. The foregoing representation and warranties set forth in this <u>Section 3.3(k)</u> shall not apply to the Intellectual Property, which is dealt with exclusively in <u>Section 3.3(q)</u>.
- (1) <u>Contracts</u>. All Material Contracts are binding upon the Company, and, to the Company's knowledge, the other parties thereto. No material default by the Company has occurred thereunder and, to the Company's knowledge, no material default by the other contracting parties has occurred thereunder.

- (m) Permits. The Company possesses all Permits that are required for the Company to conduct the Business as presently conducted and which, if not possessed, would have a Material Adverse Effect.
- (n) Employees. Section 3.3(n) of the Disclosure Schedule contains a true and correct list of all employees of the Company as of the date of this Agreement, together with their respective base salaries, expected bonuses (mandatory and discretionary) and positions. No employees have been laid off by the Company in the ninety (90) days preceding the date hereof.
- (o) <u>Litigation; Orders</u>. There are no lawsuits, actions, proceedings, investigations, claims, complaints, injunctions or orders by or before any Governmental Authority, pending or, to the Company's knowledge, threatened in writing against the Company or any of the Company's officers, directors or Affiliates, with respect to the Company's operations, Business or assets, or with respect to the consummation of the transactions contemplated hereby. The Company is not a party to, or bound by, any decree, order or arbitration award (or agreement entered into in any administrative, judicial or arbitration proceeding with any Governmental Authority) with respect to the Company's operations, Business or assets.
- (p) <u>Compliance with Laws</u>. The Company is not in violation of, or delinquent in respect to, any decree, order or arbitration award or law, statute, or regulation of or agreement with, or any Permit from, any Federal, state or local Governmental Authority to which the property, assets, personnel or Business activities of the Company are subject, in each case, which violation or delinquency would have a Material Adverse Effect.

(q) Intellectual Property.

- (i) Section 3.3(q) of the Disclosure Schedule sets forth a complete and accurate list of all the following that are owned by the Company: (1) patented or registered Intellectual Property and pending patent applications and other applications for registration of Intellectual Property; (2) all material unregistered trademarks, service marks and domain names; (3) all material Software (other than commercially-available, off-the-shelf Software); and (4) all material licenses or similar agreements or arrangements to which the Company is a licensee or licensor of Intellectual Property (excluding licenses for commercially-available, off-the-shelf Software).
- (ii) Except as set forth in Section 3.3(q) of the Disclosure Schedule, to the Company's knowledge, the Company is the owner of, or has rights to use, all of the material Intellectual Property necessary for the conduct of the Business as currently conducted (the "Company IP").
- (iii) Except as set forth on Section 3.3(q) of the Disclosure Schedule, the Company is not obligated under any Intellectual Property license agreement or otherwise to pay royalty or license fees or honorarium for the use of any Company IP.
- (r) <u>Bank Accounts</u>. <u>Section 3.3(r)</u> of the Disclosure Schedule contains a list showing: (i) the name of each bank, safe deposit company or other financial institution in which the Company has an account, lock box or safe deposit box; and (ii) the names of all Persons authorized to draw thereon or to have access thereto.

- (s) <u>Brokers</u>. Neither Seller, any of its Affiliates, nor the Company have dealt with any Person who is entitled to a broker's commission, finder's fee, investment banker's fee or similar payment from Buyer or the Company for arranging the transactions contemplated hereby or introducing the parties to each other.
 - 3.4 Representations and Warranties of Seller. Seller represents and warrants to Buyer as of the date hereof as follows:
 - (a) Organization, Existence and Good Standing. Seller is a corporation duly incorporated, existing and in good standing under the laws of the State of Nevada.
- (b) <u>Power and Authority</u>. Seller has full power and authority to execute and perform this Agreement and the other Transaction Agreements. The execution and delivery of this Agreement and the other Transaction Agreements to which Seller is a party by Seller and the performance by it of all of its obligations under this Agreement and the other Transaction Agreements to which it is a party have been duly approved prior to the date of this Agreement by all requisite action of its board of directors.
- (c) Enforceability. Each of this Agreement and the other Transaction Agreements to which it is a party has been duly executed and delivered by Seller and constitutes a legal, valid and binding agreement of Seller, enforceable against Seller in accordance with its terms, except to the extent that enforcement may be affected by laws relating to bankruptcy, reorganization, insolvency and creditors' rights and by the availability of injunctive relief, specific performance and other equitable remedies.
- (d) <u>Consents</u>. No consent, authorization, order or approval of, or filing or registration with, any Governmental Authority is required for or in connection with the consummation by Seller of the transactions contemplated hereby.
- (e) <u>Conflicts Under Constituent Documents or Laws</u>. Neither the execution and delivery of this Agreement or the other Transaction Agreements to which it is a party by Seller, nor the consummation by it of the transactions contemplated hereby will conflict with or constitute a breach of any of the terms, conditions or provisions of its certificate of incorporation or by-laws. Neither the execution and delivery of this Agreement by Seller, nor the consummation by it of the transactions contemplated hereby will conflict with or constitute a breach of any of the terms, conditions or provisions of any statute or administrative regulation, or of any order, writ, injunction, judgment or decree of any court or Governmental Authority or of any arbitration award, to which Seller is a party or by which Seller or any of Seller's assets is bound.
- (f) <u>Conflicts Under Contracts</u>. Seller is not a party to, or bound by, any unexpired, undischarged or unsatisfied written or oral contract, agreement, indenture, mortgage, debenture, note or other instruments under the terms of which the execution, delivery and performance by Seller of this Agreement and the consummation of the transactions contemplated hereby by Seller will require a consent, approval, or notice or result in a Lien on the Interest owned by Seller.
 - (g) Title. Seller owns all of the Interest, free and clear of all Liens.

- 3.5 Limitation on Warranties. The representations and warranties of the Company and Seller in Sections 3.3 and 3.4 constitute the sole and exclusive representations and warranties to Buyer in connection with the transactions contemplated hereby. Except as expressly set forth in Sections 3.3 and 3.4, neither Seller, the Company, their respective Affiliates nor any of their respective directors, managers, partners, shareholders, members, officers, employees, accounting firms, legal counsel or other agents, consultants or representatives make any express or implied representation or warranty of any kind whatsoever (including, without limitation, any representation or warranty as to the physical condition or value of any of the assets of the Company or the Business or, the future profitability or future earnings performance of the Business), and Seller and the Company (on behalf of themselves and each of their respective Affiliates and their respective directors, managers, partners, shareholders, members, officers, employees, accounting firms, legal counsel, agents, consultants and representatives) disclaim all liability and responsibility for, and Buyer acknowledges and agrees that it has not relied on, any representation, warranty, covenant, agreement, or statement made or information communicated (orally or in writing) to Buyer (including any opinion, information, or advice which may have been provided to Buyer or any of its Affiliates, directors, managers, partners, shareholders, members, officers, employees, accounting firms, legal counsel or other agents, consultants or representatives by any directors, managers, partners, shareholders, members, officers, employees, accounting firms, legal counsel or other agents, consultants or representatives by any directors, managers, partners, shareholders, members, officers, employees, accounting firms, legal counsel or other agents, consultants or representatives by any directors, managers, partners, shareholders, members, officers, employees, accounting firms, legal c
- 3.6 <u>Definition of Knowledge</u>. For the purposes of this Agreement, Company's knowledge (and words of similar import) shall be deemed to be limited to the actual knowledge of Nadir Ali, Wendy Loundermon and Shirish Tangirala, without giving effect to imputed or constructive knowledge or giving rise to any duty to investigate.

ARTICLE IV Conduct Prior to the Closing

- 4.1 General. The Company, Seller and Buyer have the rights and obligations with respect to the period between the date hereof and the earlier of the Closing and termination of this Agreement in accordance with Section 9.2, at which time the provisions of this Article IV shall terminate.
 - 4.2 Company's and Seller's Obligations. The following are the Company's and Seller's obligations (as applicable):
- (a) The Company shall give to Buyer's officers, employees, agents, attorneys, consultants, accountants and lenders reasonable access during normal business hours to all of the properties, books, contracts, documents, insurance policies, records and personnel of or with respect to the Company and shall furnish to Buyer and such Persons as Buyer shall designate to Seller such information as Buyer or such Persons may at any time and from time to time reasonably request, all in a manner so as to not interfere with the normal business operations of the Company. Notwithstanding anything to the contrary set forth in this Agreement, during the period between the date hereof and the Closing Date, neither the Company nor Seller shall in any event be required to disclose to Buyer or any of its representatives, or otherwise provide Buyer or any of its representatives any access to, any information, properties or personnel (i) if doing so would violate any contract or law to which Seller or the Company is a party or is subject or which it determines could result in a loss of the ability to successfully assert attorney-client and work product privileges, or (ii) if Seller determines that such disclosure or access should not be made or provided due to the competitively sensitive nature thereof.

- (b) The Company shall use commercially reasonable efforts to carry on the Business in the usual and ordinary course of business, consistent with past practices, except as permitted or contemplated by this Agreement or the other Transaction Agreements. Notwithstanding the forgoing, nothing contained in this Agreement shall prohibit the Company, whether or not in the usual and ordinary course of business and whether or not consistent with past practice, to pay or prepay any obligation or to pay, transfer or distribute cash to Seller.
- (c) Without limiting the generality of any other provision of this Agreement, except as (i) otherwise consented to by Buyer in writing (which consent shall not be unreasonably withheld, conditioned or delayed), (ii) otherwise contemplated by this Agreement or any other Transaction Agreement, or (iii) would constitute a violation of applicable law, from the date hereof through the Closing Date, the Company shall not, and Seller shall cause the Company not to:
 - (i) amend the certificate of formation or limited liability company agreement of the Company except as contemplated herein;
 - (ii) make any change in the Company's capitalization or issue any security of any class or issue or become a party to any subscriptions, warrants, rights, options, convertible securities or other agreements or commitments of any character relating to the limited liability company interests of the Company;
 - (iii) pay or declare any dividend or make any distribution on its limited liability company interests, or purchase or redeem any of its limited liability company interests, other than (1) payments expressly permitted in <u>Section 4.2(b)</u> and (2) as otherwise contemplated hereunder;
 - (iv) materially increase the compensation payable to any employee, except in the ordinary course of business consistent with past practices;
 - (v) make any payment to its employees, officers or managers except such amounts as constitute currently effective compensation for services rendered or reimbursement for out-of-pocket business expenses and except as otherwise contemplated hereunder;
 - (vi) institute or amend any employee benefit program or fringe benefit program with respect to the employees of the Company; or

(vii) sell, transfer or otherwise dispose of any asset or property (other than as expressly permitted hereunder), except for sales of inventory and for transfers of cash in payment of the Company's liabilities, all in the usual and ordinary course of business in accordance with past practices.

4.3 <u>Buyer's Obligations</u>. Buyer's obligations are as follows:

- (a) Buyer shall, at its cost and expense, make all filings with, provide all notices to, and obtain all Permits from, Governmental Authorities as are necessary to be made or provided by Buyer and/or its Affiliates in connection with the transactions contemplated by this Agreement, including, without limitation, all filings with, all notices to, and all Permits from applicable Governmental Authorities having regulatory control and/or jurisdiction with respect to the Business.
- (b) Except as required by this Agreement, Buyer and its Affiliates shall not, without the prior written consent of the Company, engage in any action or enter into any transaction or permit any action to be taken or transaction to be entered into, that would materially impair or delay Buyer's ability to consummate the transactions contemplated hereby or perform their obligations hereunder.
 - 4.4 <u>Joint Obligations</u>. The following shall apply with equal force to the Company, Seller and Buyer:
- (a) Subject to the terms of this Agreement, each of the parties shall use their best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable to consummate the transactions contemplated hereby as soon as practicable.
- (b) Each party shall promptly give each other party written notice of the existence or occurrence of any condition that (i) would make any representation or warranty herein contained of either party untrue, other than changes which would not be reasonably expected to have a Material Adverse Effect, or (ii) might reasonably be expected to prevent the timely consummation of the transactions contemplated hereby.
- (c) No party shall intentionally perform any act that, if performed, or intentionally omit to perform any act that, if omitted to be performed, would prevent or excuse the performance of this Agreement by any party or which would result in any representation or warranty herein contained of such party being untrue in any material respect as if originally made on and as of the Closing Date, other than changes which would not be reasonably expected to have a Material Adverse Effect.

ARTICLE V Conditions to Closing

- 5.1 <u>Conditions to the Company's and Seller's Obligations</u>. The obligation of the Company and Seller to consummate the transactions contemplated hereby is subject to the fulfillment of all of the following conditions at or prior to the Closing and as of the Closing, upon the non-fulfillment of any of which this Agreement may, at the Company's option, be terminated if permitted pursuant to and with the effect set forth in <u>Article IX</u>:
- (a) The representations and warranties made by Buyer set forth in Section 3.2 shall be true and correct in all material respects as if originally made on and as of the Closing Date (or, if made as of a specific date in the text of such representations and warranties, at and as of such date), except as affected by the transactions contemplated by this Agreement and except for such failures of representations or warranties to be true and correct which have not had a material adverse effect on the ability of Buyer to consummate the transactions contemplated by this Agreement;
- (b) Buyer shall have performed in all material respects all the covenants and agreements required to be performed by it under this Agreement at or prior to the Closing;
- (c) Buyer shall have delivered to the Company and Seller a certificate dated the Closing Date and signed by an authorized representative of Buyer on behalf of Buyer certifying that the conditions specified in <u>Sections 5.1(a)</u> and <u>5.1(b)</u> have been satisfied (the "*Buyer Closing Certificate*");
- (d) No lawsuit, proceeding or investigation shall have been commenced by any Governmental Authority on any grounds to restrain, enjoin or hinder the consummation of the transactions contemplated hereby.
- 5.2 <u>Conditions to Buyer's Obligations</u>. The obligation of Buyer to consummate the transactions contemplated hereby is subject to the fulfillment of all of the following conditions at or prior to the Closing and as of the Closing, upon the non-fulfillment of any of which this Agreement may, at Buyer's option, be terminated if permitted pursuant to and with the effect set forth in <u>Article IX</u>:
- (a) The representations and warranties made by the Company and the Seller set forth in Section 3.3 and Section 3.4 shall be true and correct in all material respects as if originally made on and as of the Closing Date (or, if made as of a specific date in the text of such representations and warranties, as of such date), except as affected by the transactions contemplated by this Agreement and the other Transaction Agreements and except where the failure of such representations and warranties to be true and correct would not have a Material Adverse Effect;
- (b) The Company and Seller shall have performed in all material respects all the covenants and agreements required to be performed by it or them under this Agreement at or prior to the Closing;
- (c) The Company and the Seller shall have delivered to Buyer a certificate dated the Closing Date and signed by an authorized representative of the Company on behalf of the Company and the Seller on behalf of the Seller certifying that the conditions specified in Sections 5.2(a) and 5.2(b) have been satisfied (the "Company Closing Certificate");

- (d) During the period from the date of this Agreement to the Closing Date, there shall not have occurred any event which has resulted in a Material Adverse Effect; and
- (e) No lawsuit, proceeding or investigation shall have been commenced by any Governmental Authority on any grounds to restrain, enjoin or hinder the consummation of the transactions contemplated hereby; and
 - (f) The transactions contemplated by the Contribution Agreement shall have been consummated.
- 5.3 <u>Frustration of Closing Conditions</u>. No party hereto may rely on the failure of any condition set forth in <u>Sections 5.1</u> or <u>5.2</u>, as the case may be, if such failure was caused by such party's failure to comply with any provision of this Agreement.
 - 5.4 Waiver of Conditions. All conditions set forth in this Article V will be deemed to have been satisfied or waived from and after the Closing.

ARTICLE VI Closing

- 6.1 Form of Documents. At the Closing, the parties shall deliver the documents, and shall perform the acts that are set forth in this Article VI. All documents that the Company and Seller shall deliver shall be in form and substance reasonably satisfactory to Buyer and Buyer's counsel. All documents that Buyer shall deliver shall be in form and substance reasonably satisfactory to Seller and Seller's counsel.
- 6.2 <u>Buyer's Deliveries</u>. Buyer shall execute and/or deliver to Seller all of the documents reasonably required from Buyer to consummate the transactions contemplated hereby.
 - 6.3 Seller's Deliveries. Seller shall execute or deliver to Buyer all of the following:
 - (a) an assignment of the Purchased Interest, duly executed by Seller;
 - (b) a certified copy of the Company's certificate of formation issued by the Secretary of State of the State of Nevada;
 - (c) a certificate of good standing of the Company issued by the Secretary of State of the State of Nevada;
 - (d) a form W-9; and
 - (e) the Transition Services Agreement; and
- (f) a consent, duly executed by Seller as managing member, appointing Nadir Ali as Chief Executive Officer of the Company and Shirish Tangirala as Vice President of the Company; and
- (g) without limitation by specific enumeration of the foregoing, all other documents reasonably required from Seller to consummate the transactions contemplated hereby.

6.4 Company's Deliverables. Company shall execute or deliver to Buyer all of the following:

- (a) an amendment to the Company's Operating Agreement, pursuant to which Buyer shall be identified as the sole managing member of Company; and
- (b) without limitation by specific enumeration of the foregoing, all other documents reasonably required from Company to consummate the transactions contemplated hereby.

ARTICLE VII Post-Closing Agreements

- 7.1 Post-Closing Agreements. From and after the Closing, the parties shall have the respective rights and obligations that are set forth in the remainder of this Article VII.
- 7.2 <u>Inspection of Records</u>. From and after the Closing, Buyer shall, and shall cause the Company to, provide Seller and its agents with reasonable access (for the purpose of examining and copying), during normal business hours, and upon reasonable advance notice, to the books and records of the Company with respect to periods or occurrences prior to the Closing and reasonable access, during normal business hours, and upon reasonable advance notice, to employees of each of Buyer, the Company, and each of their respective Affiliates for purposes of complying with any applicable Tax, financial reporting or regulatory requirements, or any other reasonable business purpose. Unless otherwise consented to in writing by Seller, neither Buyer nor the Company shall, for a period of two (2) years following the payment of the Second Installment, destroy, alter or otherwise dispose of any of the books and records of the Company for any period prior to the Closing without first offering to surrender to Seller such books and records or any portion thereof which Buyer or the Company may intend to destroy, alter or dispose of.

7.3 Intentionally Omitted.

7.4 Tax Matters.

(a) Buyer shall prepare or cause to be prepared and timely file or cause to be timely filed all Tax Returns of the Company and its subsidiaries with respect to any Straddle Periods and any Pre-Closing Tax Periods where the applicable Tax Return is not due (taking extensions into account) until after the Closing Date (each, a "Buyer" Prepared Tax Return"). Any such Buyer Prepared Tax Return shall be prepared consistent with past practices, unless otherwise required by applicable law. Buyer shall provide Seller with a copy of any such Buyer Prepared Tax Return that is an income Tax Return (each, a "Buyer Income Tax Return") for Seller's review, comment and approval no later than thirty (30) days prior to its due date (taking into account any extensions). Seller shall review and comment on such Buyer Income Tax Returns within fifteen (15) days following receipt thereof. If Seller does not submit comments within such period, then Seller shall be deemed to have approved such Buyer Income Tax Returns as prepared by Buyer. If Seller objects to any item on such Buyer Income Tax Return, it shall, within fifteen (15) days after delivery of such Buyer Income Tax Return, notify Buyer in writing that it so objects. Upon a timely delivery of a notice of objection, Buyer and Seller shall negotiate in good faith to resolve such items. If the parties are unable to resolve any such dispute within five (5) days after receipt by Buyer of such notice, the disputed items shall be referred to an independent accounting firm of international or national standing jointly selected by Buyer and Seller (the "Independent Accountant"), acting as an expert and not an arbitrator, for resolution on at least a more-likely-than-not basis, and any determination by the Independent Accountant shall be final and binding. The Independent Accountant shall resolve any disputed items within ten (10) days of having the item referred to it pursuant to such procedures as it may require. Upon the final determination of such dispute, Buyer shall file or cause to be filed any Buyer Income Tax Return (including, if necessary, as an amended Tax Return) promptly but no later than five (5) days after such final determination; provided that if the Independent Accountant cannot resolve a dispute (or if Seller and Buyer are unable to resolve any disagreements) before the due date for filing the Buyer Income Tax Return, Buyer shall timely file the Buyer Income Tax Return reflecting any changes to which Buyer and Seller have agreed and shall be permitted to file an amendment to such Buyer Income Tax Return if the Independent Accountant determines that such amendment is required. The costs, fees and expenses of the Independent Accountant shall be borne equally by Buyer and Seller. Buyer shall provide a copy of each Buyer Prepared Tax Return to Seller promptly after it is filed.

- (b) Buyer and Seller shall, and Buyer shall cause the Company and its subsidiaries to, reasonably cooperate fully, as and to the extent reasonably requested, in connection with the preparation and filing of Tax Returns and any audit, litigation or other proceeding with respect to Taxes and Tax Returns of or with respect to the Company or any of its subsidiaries for any Pre-Closing Tax Period. Such cooperation shall include the retention, and (upon the other party's request) the provision, of records and information which are reasonably relevant to any such Tax Return, audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder; provided that the party requesting assistance shall pay the reasonable out-of-pocket expenses incurred by the party providing such assistance.
- (c) Buyer shall not (and Buyer shall cause each of its Affiliates (including the Company and its subsidiaries after the Closing) to not) without the prior written consent of the Seller: (i) file, amend, re-file or otherwise modify any Tax Return relating in whole or in part to the Company or any of its subsidiaries with respect to any Pre-Closing Tax Period or Straddle Period, (ii) extend or waive the applicable statute of limitations with respect to a Tax or Tax Return of the Company or any of its subsidiaries for any Pre-Closing Tax Period or Straddle Period, (iii) file any ruling or request with any taxing authority that primarily relates to Taxes or Tax Returns of the Company or any of its subsidiaries for a Pre-Closing Tax Period or Straddle Period, (iv) make or change any election or method of accounting with respect to, or that has retroactive effect to, any Pre-Closing Tax Period or Straddle Period of the Company or any of its subsidiaries, (v) make any voluntary disclosure to, or otherwise voluntarily approach or initiate contact with, any Governmental Authority with respect to any Tax or Tax Return of the Company or any of its subsidiaries for any Pre-Closing Tax Period or Straddle Period, or (vi) cause or permit the Company or any of its subsidiaries to take any action outside the ordinary course of business on the Closing Date after the Closing.
- (d) For U.S. federal and applicable state income Tax purposes, Buyer and Seller intend and agree that (i) the purchase and sale of the Purchased Interest shall be treated as the purchase and sale of all of the assets of the Company, subject to the liabilities of the Company, and (ii) following Closing the Buyer and Seller shall be treated as independent parties and they shall not be treated as partners in a partnership, joint venturers or otherwise in a joint undertaking. Buyer and Seller shall prepare and file, and they shall cause their respective Affiliates to prepare and file, all U.S. federal and state income Tax Returns consistent with such intended Tax treatment. Buyer and Seller agree to allocate the Purchase Price (plus all other amounts that are treated as consideration for U.S. federal income Tax purposes) among the assets of the Company in accordance with the methodology set forth on Schedule 7.4(d) (the "Allocation Methodology"). Within ninety (90) days after the Working Capital Adjustment is finally determined, Seller shall deliver to Buyer an allocation of the Purchase Price (plus all other amounts that are treated as consideration for U.S. federal income Tax purposes) among the assets of the Company in accordance with the Allocation Methodology (the "Allocation Schedule"). The parties agree to prepare and file all U.S. federal and applicable state income Tax Returns in a manner consistent with the Allocation Schedule. Any subsequent adjustments to the Purchase Price shall be allocated in accordance with the Allocation Methodology.

- (e) Buyer and Seller shall promptly provide each other with notice of any Tax inquiries, audits, examinations, proceedings or proposed Tax adjustments or assessments by any Governmental Authority that relate to any Tax or Tax Return of the Company or any of its subsidiaries for any Pre-Closing Tax Period or any Straddle Period (each, a "Tax Contest"). Buyer shall have the right to control all proceedings with respect to Tax Contests. With respect to any Tax Contest: (i) the Buyer shall keep the Seller reasonably informed regarding the status of the Tax Contest, including by promptly providing to the Seller copies of any and all correspondence received from or provided to the applicable Governmental Authority related to such Tax Contest, (ii) the Seller shall have the right to participate in such Tax Contest, including by attending meetings or conferences with the applicable Governmental Authority, (iii) the Buyer shall give the Seller the opportunity to review and comment on all submissions or filing with the applicable Governmental Authority before such submissions and filings are provided to the Governmental Authority, and (iv) the Buyer shall not settle, resolve, compromise or abandon (and shall not allow the Company or any of its subsidiaries to settle, resolve, or abandon) such Tax Contest without the prior written consent of the Seller (which shall not be unreasonably withheld, conditioned or delayed).
- 7.5 Employee Matters. Buyer will make commercially reasonable efforts to ensure that each rank and file employee of the Company who continues employment with Buyer or the Company, as applicable, after the Closing Date (a "Continuing Employee") will, during the period commencing at the Closing and ending on the first anniversary of the Closing, be provided with a rate of base salary or wages and bonus or commission opportunity that is not less favorable than the rate of base salary or wages and bonus or commission opportunity paid by the Company to such Continuing Employee immediately prior to the Closing.
- 7.6 <u>Further Assurances</u>. The parties shall execute such further documents, and perform such further acts, as may be necessary to transfer and convey the Purchased Interest to Buyer on the terms herein contained, and to otherwise comply with the terms of this Agreement and consummate the transactions contemplated hereby.

ARTICLE VIII Survival; Remedies

- 8.1 <u>Survival</u>. All representations and warranties of the Company and Seller contained in, or arising out of, this Agreement or any other Transaction Agreement (or any certificate or other documents delivered in connection herewith or therewith) shall expire and terminate at the consummation of the transactions contemplated hereby at the Closing, and thereafter no claim may be made with respect to, or any suit or other proceeding instituted for, any breach of or inaccuracy in any such representation or warranty (and consummation of the Closing shall be deemed a waiver of any and all breaches of or inaccuracies in any such representation and warranty and all of Buyer's rights and remedies with respect thereto). All covenants and agreements contained herein to be performed prior to or at the Closing shall expire and terminate at Closing, and thereafter no claim may be made with respect to, or any suit or other proceeding instituted for, any breach of or failure to perform any such covenant or agreement (and consummation of the Closing shall be deemed a waiver of any and all breaches of or failures to perform any such covenant or agreement and all of Buyer's rights and remedies with respect thereto). All covenants and agreements contained in this Agreement or any other Transaction Agreement, in each case, which by their terms are to be performed (or which prohibit actions) subsequent to the Closing Date ("Post-Closing Covenants") will survive the Closing for the period expressly specified therein, and thereafter no claim may be made with respect to, or any suit or other proceeding instituted for, any breach of or failure to perform any Post-Closing Covenants.
 - 8.2 Remedies. Notwithstanding anything expressed or implied herein to the contrary:
- (a) Buyer acknowledges and agrees that Seller shall not have any direct or indirect liability (derivatively or otherwise) with respect to any breach of or inaccuracy in any representation or warranty contained in, or arising out of, this Agreement or any other Transaction Agreement (or any certificate or other documents delivered in connection herewith), and that no claim may be made, or any suit or other proceeding instituted, by Buyer against Seller with respect thereto; and
- (b) Except with respect to actions seeking injunctive relief and/or specific performance, and except as set forth in Section 2.5, from and after the Closing, the sole and exclusive remedy of the parties with respect to any matters arising under or relating to this Agreement, the other Transaction Agreements and/or the transactions contemplated hereby and thereby shall be a contract action to enforce the terms of the Post-Closing Covenants.

ARTICLE IX Effect of Termination/Proceeding

9.1 General. The parties shall have the rights and remedies with respect to the termination and/or enforcement of this Agreement that are set forth in this Article IX.

- 9.2 <u>Right to Terminate</u>. Anything to the contrary herein notwithstanding, this Agreement and the transactions contemplated hereby may be terminated at any time prior to the Closing:
 - (a) by the mutual written consent of Buyer and Seller;
- (b) by prompt notice given in accordance with Section 10.2, by Buyer or Seller if the Closing shall not have occurred at or before 11:59 p.m. (Pacific time) on the earlier to occur of (i) the Termination Date (as defined in the XTI Merger Agreement), or (ii) the 30 Business Day following the Closing Date (as defined in the XTI Merger Agreement) (the "Termination Date"); provided that the right to terminate this Agreement under this Section 9.2(b) shall not be available to any party whose failure to fulfill any of its obligations under this Agreement has been the cause of, or resulted in the failure of, the Closing to occur on or prior to the Termination Date;
- (c) by Buyer, if there has been a breach of any representation, warranty or covenant made by Seller or the Company in this Agreement, such that any condition set forth in Section 5.2 is not capable of being satisfied prior to the Termination Date and Buyer has not waived such condition; provided that the right to terminate this Agreement pursuant to this Section 9.2(c) shall not be available to Buyer if the failure of Buyer to fulfill any of its obligations under this Agreement has been the cause of, or resulted in, such breach;
- (d) by Seller, if there has been a breach of any representation, warranty or covenant made by Buyer in this Agreement, such that any condition set forth in Section 5.1 is not capable of being satisfied prior to the Termination Date and Seller has not waived such condition; provided that the right to terminate this Agreement pursuant to this Section 9.2(d) shall not be available to Seller if the failure of Seller or the Company to fulfill any of their obligations under this Agreement has been the cause of, or resulted in, such breach; or
- (e) by either Buyer or Seller, in writing, if any Governmental Authority shall have entered any injunction, order, decree or ruling which, in any such case, has become final and non-appealable and has the effect of preventing or prohibiting consummation of the transactions contemplated by this Agreement; <u>provided</u> that the provisions of this <u>Section 9.2(e)</u> shall not be available to any party unless such party shall have used its reasonable best efforts to oppose any such action or to have such action vacated or made inapplicable to the transactions contemplated by this Agreement.
- 9.3 <u>Certain Effects of Termination</u>. In the event of the termination of this Agreement by either Seller or Buyer as provided in <u>Section 9.2</u> each party, if so requested by the other party, will return promptly every document furnished to it by the other party (or any subsidiary, division, associate or Affiliate of such other party) in connection with the transactions contemplated hereby, whether so obtained before or after the execution of this Agreement, and any copies thereof (except for copies of documents publicly available) that may have been made, and will use reasonable efforts to cause its representatives and any representatives of financial institutions and investors and others to whom such documents were furnished promptly to return such documents and any copies thereof any of them may have made. This <u>Section 9.3</u> shall survive any termination of this Agreement.

- 9.4 Remedies. Notwithstanding any termination right granted in Section 9.2, in the event of the non-fulfillment of any condition to a party's closing obligations, in the alternative, such party may elect to do one of the following:
- (a) proceed to close despite the non-fulfillment of any closing condition (if legally permissible), it being understood that consummation of the Closing shall be deemed a waiver of a breach of any representation, warranty or covenant and of such party's rights and remedies with respect thereto to the extent that such party shall have actual knowledge of such breach and the Closing shall nonetheless occur;
 - (b) decline to close, terminate this Agreement as provided in Section 9.2, and thereafter seek damages to the extent permitted in Section 9.5; or
- (c) seek specific performance by the other party hereto of such other party's obligations hereunder which it has failed to perform so that Closing may proceed (it being acknowledged and agreed that the non-breaching party would be damaged irreparably, the remedies available at law to the non-breaching party would be inadequate, and the performance of such other party's obligations under this Agreement may be specifically enforced).
- 9.5 Right to Damages. If this Agreement is terminated pursuant to Section 9.2, no party hereto shall have any claim for monetary damages against the other, except if the circumstances giving rise to such termination were caused by the other party's willful failure to comply with a material covenant set forth herein, in which event termination pursuant to Section 9.2 shall not be deemed or construed as limiting or denying any legal or equitable right or remedy of such party, and such party shall also be entitled to recover its costs and expenses which are incurred in pursuing its rights and remedies (including reasonable attorneys' fees).

ARTICLE X Miscellaneous

- 10.1 Publicity. Except as otherwise required by law or applicable stock exchange rules, press releases and other publicity concerning this transaction shall be made only with the prior agreement of Seller and Buyer (and in any event, the parties shall use all reasonable efforts to consult and agree with each other with respect to the content of any such required press release or other publicity). Except as otherwise required by law or applicable stock exchange rules, no such press releases or other publicity shall state the amount of the Purchase Price.
- 10.2 Notices. All notices, requests, demands and other communications under this Agreement shall be in writing and shall be deemed to have been duly given: (a) on the day of service if served personally on the party to whom notice is to be given; (b) on the day of transmission if delivered by electronic mail during regular business hours on a Business Day and, if not, then on the following Business Day; or (c) one (1) Business Day after being sent by Fed Ex or a similar nationally recognized overnight courier (with next day delivery specified):

If to Seller:

Inpixon 405 Waverley Street Palo Alto, California 94301 Attention: Melanie Figueroa, General Counsel

Email: melanie.figueroa@inpixon.com

with a copy to (which shall not constitute notice):

Norton Rose Fulbright US LLP 1045 West Fulton Market, Suite 1200 Chicago, IL 60607 Attention: Kevin Friedmann Email: kevin.friedmann@nortonrosefulbright.com

If to the Company (prior to the Closing):

c/o Inpixon
Grafiti LLC
405 Waverley Street
Palo Alto, California 94301
Attention: Nadir Ali, Chief Executive Officer
Email: nadir.ali@inpixon.com

with a copy to (which shall not constitute notice):

Norton Rose Fulbright US LLP 1045 West Fulton Market, Suite 1200 Chicago, IL 60607 Attention: Kevin Friedmann Email: kevin.friedmann@nortonrosefulbright.com

If to Buyer:

Grafiti Group LLC 405 Waverley Street Palo Alto, California 94301 Attention: Nadir Ali Email: nadir.ali@grafiti.com_

Any party may change its address for the purpose of this Section 10.2 by giving the other party written notice of its new address in the manner set forth above.

10.3 Expenses; Transfer Taxes. Except as otherwise expressly set forth herein, each party hereto shall bear all fees and expenses incurred by such party in connection with, relating to or arising out of the negotiation, preparation, execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby, including financial advisors', attorneys', accountants' and other professional fees and expenses, whether or not the transactions contemplated by this Agreement are consummated, provided, however, that the Seller shall bear all such fees and expenses incurred by the Seller or the Company through the Closing Date. Notwithstanding the foregoing, in the event any party institutes any legal suit, action, arbitration or other proceeding against the other party to enforce the provisions of this Agreement or otherwise related to the transactions contemplated hereunder, the prevailing party in such suit, action arbitration or proceeding shall be entitled to recover from the other party the costs and expenses incurred by the prevailing party in conducting and/or defending such suit, action, arbitration or proceeding, including reasonable attorneys' fees and other professional fees and expenses incurred in connection therewith (it being understood that this sentence shall not apply to costs and expenses incurred pursuant to an arbitration proceeding under Section 2.5, but shall apply to any action to enforce the decision of the Arbitrating Accountant thereunder). All transfer, stamp, documentary, sales and use, value added, registration, deed and other similar Taxes, and related fees imposed on any of the parties by any Governmental Authority in connection with the transactions contemplated by this Agreement ("Transfer Taxes"), shall be borne solely by Buyer, and Buyer shall timely file, and Seller will cooperate in filing, all Tax Returns related to such Taxes.

- 10.4 Entire Agreement. This Agreement and the other Transaction Agreements, constitute the entire agreement between the parties and shall be binding upon and inure to the benefit of the parties hereto and their respective legal representatives, successors and permitted assigns. Each Appendix, Attachment, Exhibit, Schedule and the Disclosure Schedule, shall be considered incorporated into this Agreement. Any amendments, or alternative or supplementary provisions, to this Agreement must be made in writing and duly executed by an authorized representative or agent of each of the parties hereto.
- 10.5 <u>Projections</u>. Neither Seller, the Company, their respective Affiliates nor any of their respective directors, managers, partners, shareholders, members, officers, employees, accounting firms, legal counsel or other agents, consultants or representatives make any express or implied warranty of any kind whatsoever, including any representation or warranty as to the future profitability or future earnings performance of the Company. Buyer acknowledges that any estimates, forecasts, or projections furnished or made available to it concerning the Company or its properties, business, assets or liabilities have not been prepared in accordance with GAAP or standards applicable under the Securities Act, and such estimates, forecasts and projections reflect numerous assumptions, and are subject to material risks and uncertainties. Buyer acknowledges that actual results may vary, perhaps materially, and Buyer is not relying on any such estimates, forecasts or projections.
- 10.6 Non-Waiver. The failure in any one or more instances of a party to insist upon performance of any of the terms, covenants or conditions of this Agreement, to exercise any right or privilege in this Agreement conferred, or the waiver by said party of any breach of any of the terms, covenants or conditions of this Agreement, shall not be construed as a subsequent waiver of any such terms, covenants, conditions, rights or privileges, but the same shall continue and remain in full force and effect as if no such forbearance or waiver had occurred. Except as expressly provided herein, no waiver shall be effective unless it is in writing and signed by an authorized representative of the waiving party.
- 10.7 <u>Counterparts</u>. This Agreement and any of the Transaction Agreements may be executed in multiple counterparts, each of which shall be deemed an original and all of which together shall constitute one instrument. This Agreement and any of the Transaction Agreements may be executed through the exchange of pdf or other electronic signature pages, which shall have the same legal effect as original signatures.

- 10.8 Severability. Whenever possible, each provision of this Agreement shall be interpreted in such a manner as to be effective and valid under applicable law, but if any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect any other provision or any other jurisdiction, and, for purposes of such jurisdiction, such provision or portion thereof shall be struck from the remainder of this Agreement, which shall remain in full force and effect. This Agreement shall be reformed, construed and enforced in such jurisdiction so as to best give effect to the intent of the parties under this Agreement.
- 10.9 <u>Applicable Law</u>. This Agreement shall be construed and enforced in accordance with, and all questions concerning the construction, validity, interpretation, inducement to enter and/or performance of this Agreement (whether related to breach of contract, tortious conduct or otherwise and whether now existing or hereafter arising) shall be governed by, the internal laws of the State of Nevada, without giving effect to any laws, rules or provisions (whether of the State of Nevada or any other jurisdiction) that would cause the application of the laws, rules or provisions of any jurisdiction other than the State of Nevada.
- 10.10 <u>Binding Effect; Benefit</u>. This Agreement shall inure to the benefit of and be binding upon the parties hereto, and their respective successors and permitted assigns. Nothing in this Agreement, express or implied, shall confer on any Person other than the parties hereto, and their respective successors and permitted assigns, any rights, remedies, obligations or liabilities under or by reason of this Agreement, including third party beneficiary rights, except that the Indemnified Employees shall be third party beneficiaries of <u>Section 7.3</u>, NRF shall be a third party beneficiary of <u>Section 10.16</u> and the Released Persons shall be third party beneficiaries of <u>Section 10.17</u>.
- 10.11 <u>Assignability</u>. Neither this Agreement nor any of the rights, interests or obligations hereunder may be assigned or delegated by any of the parties hereto without the prior written consent of Buyer and Seller.
- 10.12 Rule of Construction. The parties acknowledge and agree that each has negotiated and reviewed the terms of this Agreement, assisted by such legal and tax counsel as they desired, and has contributed to its revisions. The parties further agree that the rule of construction that any ambiguities are resolved against the drafting party will be subordinated to the principle that the terms and provisions of this Agreement will be construed fairly as to all parties and not in favor of or against any party. The terms "including", "includes", "include" and words of like import shall be construed broadly as if followed by the words "without limitation" or "but not limited to." The terms "herein", "hereunder", "hereof" and words of like import refer to this entire Agreement instead of just the provision in which they are found. The term "pending" shall mean pending (but shall not be construed as referring to any action, suit or proceeding against the Company that has been filed but not yet served on the Company), and "threatened" means threatened (and shall be construed as referring, without limitation, to any action, suit or proceeding against the Company that has been filed but not yet served on the Company). The term "extent" in the phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such phrase shall not mean simply "if". Unless otherwise expressly indicated, all dollar amounts referred to in this Agreement (and the other Transaction Agreements) are in United States Dollars, and all amounts owing under this Agreement and the other Transaction Agreements shall be paid in United States Dollars. The headings contained in this Agreement are for convenience of reference only and shall not affect the meaning or interpretation of this Agreement.

10.13 <u>WAIVER OF TRIAL BY JURY</u>. EACH OF THE PARTIES HERETO WAIVES THE RIGHT TO A JURY TRIAL IN CONNECTION WITH ANY LAWSUIT, ACTION OR PROCEEDING SEEKING ENFORCEMENT OF SUCH PARTY'S RIGHTS UNDER THIS AGREEMENT AND/OR RELATING TO THE TRANSACTIONS CONTEMPLATED HEREUNDER.

10.14 Consent to Jurisdiction. Each party hereto agrees that any lawsuit, action or other proceeding arising out of or relating to this Agreement or any transaction contemplated hereby shall be brought exclusively in the state courts of Nevada, or in the event (but only in the event) that any such court does not have subject matter jurisdiction over such lawsuit, action or other proceeding, the United States District Court for the District of Nevada, and each of the parties hereto hereby submits to the exclusive jurisdiction of such courts for itself and with respect to its property, generally and unconditionally, for the purpose of any such lawsuit, action or other proceeding. A final judgment in any such lawsuit, action or other proceeding may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Each party hereto agrees not to commence any lawsuit, action or other proceeding arising out of or relating to this Agreement or the transactions contemplated hereby except in the courts described above (other than actions in any court of competent jurisdiction to enforce any judgment, decree or award rendered by any such court in Nevada as described above), irrevocably and unconditionally waives any objection to the laying of venue of any lawsuit, action or other proceeding arising out of or relating to this Agreement or the transactions contemplated hereby in any such court, and hereby irrevocably and unconditionally waives and agrees not to plead or claim in any such court that any such lawsuit, action or other proceeding brought in any such court has been brought in an inconvenient forum or does not have jurisdiction over any party hereto.

10.15 <u>Amendments</u>. This Agreement shall not be modified or amended except pursuant to an instrument in writing executed and delivered on behalf of each of the parties hereto.

10.16 Conflicts and Privilege. Each of the parties to this Agreement hereby agrees, on its own behalf and on behalf of its directors, members, partners, officers, employees and Affiliates, that Norton Rose Fulbright US LLP ("NRF") may serve as counsel to Seller on the one hand, and the Company on the other hand, in connection with the negotiation, preparation, execution and delivery of this Agreement and the consummation of the transactions contemplated hereby, and that, following consummation of the transactions contemplated hereby, NRF may serve as counsel to Seller or any director, member, partner, officer, employee or Affiliate of Seller in connection with any litigation, claim or obligation arising out of or relating to this Agreement or the transactions contemplated by this Agreement or any other matter notwithstanding such representation (or continued representation) of the Company and each of the parties hereto hereby consents thereto and waives any conflict of interest arising therefrom, and each of such parties shall cause any Affiliate thereof to consent to waive any conflict of interest arising from such representation. Buyer further agrees that, as to all communications among NRF, the Company and Seller that relate in any way to the transactions contemplated by this Agreement, the attorney-client privilege and the expectation of client confidence belongs to Seller and may be controlled by Seller and shall not pass to or be claimed by Buyer or the Company. Notwithstanding the foregoing, in the event that a dispute arises between Buyer, the Company and a third party other than a party to this Agreement after the Closing, the Company may assert the attorney-client privilege to prevent disclosure of confidential communications by NRF to such third party; provided that the Company may not waive such privilege without the prior written consent of Seller.

10.17 Release and Indemnity. (a) Buyer, on its own behalf and on behalf of the other Buyer Indemnitors and the Company (each, together with Buyer, a "Releasing Person") collectively, the "Releasing Persons") hereby unconditionally and irrevocably and forever releases and discharges (and, upon request from Seller, Buyer shall cause each other Releasing Person to acknowledge and agree in writing to such release and discharge) each of the officers and managers of the Company, the Seller Indemnitees and their respective Affiliates, successors and assigns, and all of their respective current and former officers, directors, managers, shareholders, members, partners, employees, agents and representatives (each, a "Released Person") from, and the Buyer Indemnitors, jointly and severally, hereby indemnity the Seller Indemnitees against, all debts, demands, causes of action, suits, covenants, torts, damages and any and all claims, defenses, offsets, judgments, demands and liabilities whatsoever, of every name and nature, both at law and in equity, known or unknown, suspected or unsuspected, accrued or unaccrued, which have been or could have been asserted against any Released Person, which any Releasing Person ever had, now has or ever may have or claim to have, which arise out of or in any way relate to events, circumstances or actions occurring, existing or taken prior to or as of the Closing Date in respect of matters relating to the Business and the Company (collectively, the "Released Claims"); provided that the parties acknowledge and agree that this Section 10.17 does not apply to and shall not constitute a release of any rights or obligations arising under this Agreement or any of the other Transaction Agreements. Buyer, on behalf of itself and each of the other Releasing Persons, covenants that none of the Releasing Persons will (and that Buyer will cause all other Persons who may seek to claim as, by, through or in relation to any of the Released Persons on the basis of or in any way relating to any of the Rele

(b) <u>Acknowledgment of Unknown Losses or Claims</u>. Buyer, on its own behalf and on behalf of each other Releasing Person, expressly understands and acknowledges that it is possible that unknown losses or claims exist or might come to exist or that present losses may have been underestimated in amount, severity, or both. Accordingly, Buyer and each other Releasing Person are deemed expressly to understand provisions and principles of law such as Section 1542 of the Civil Code of the State of California ("Section 1542") (as well as any and all provisions, rights and benefits conferred by any law of any state or territory of the United States, or principle of common law, which is similar or comparable to Section 1542), which provides: A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE, AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASING PARTY. Buyer and each other Releasing Person are hereby deemed to agree that the provisions of Section 1542 and all similar federal or state laws, rights, rules, or legal principles of California or any other jurisdiction that may be applicable herein, are hereby knowingly and voluntarily waived and relinquished with respect to the releases in Section 10.17(a).

(Signature page follows)

IN WITNESS WHEREOF, the parties have executed this Equity Purchase Agreement as of the date first above written.

SELLER:

INPIXON

By: /s/ Wendy Loundermon
Name: Wendy Loundermon
Title: Chief Financial Officer

COMPANY:

GRAFITI LLC

Inpixon, its Managing Member

By: /s/ Wendy Loundermon
Name: Wendy Loundermon
Title: Chief Financial Officer

BUYER:

GRAFITI GROUP LLC

By: /s/ Nadir Ali
Name: Nadir Ali
Title: Managing Member

Schedule 7.4(d) Allocation Methodology

Asset Class Pursuant to	
Treasury Regulation section 1.338-6(b)	Allocated Amount
Class I	Actual amount, if any, of the Class I assets included in the final calculation of the Working Capital Adjustment.
Class II	Zero
Class III	Actual amount, if any, of the Class III assets included in the final calculation of the Working Capital Adjustment.
Class IV	An amount equal to the net book value of the Class IV assets on the Closing Balance Sheet.
Class V	An amount equal to the net book value of the Class V assets on the Closing Balance Sheet.
Class VI and Class VII	The remaining balance.

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

The following unaudited pro forma condensed combined financial information gives effect to the 1-100 Inpixon Reverse Stock Split that will go effective before the close of the transaction.

The following unaudited pro forma condensed combined financial information presents the combination of the financial information of Inpixon and Subsidiaries (the "Company" or "Inpixon") and Legacy XTI (defined below) adjusted to give effect to the reverse Merger and related transactions. The following unaudited pro forma condensed combined financial information has been prepared in accordance with Article 11 of Regulation S-X. Defined terms included below have the same meaning as terms defined and included elsewhere in this Form 8-K.

The historical financial information of Legacy XTI was derived from the unaudited financial statements of XTI Aircraft Company ("Legacy XTI") as of and for the nine months ended September 30, 2023 and the audited financial statements of Legacy XTI for the year ended December 31, 2022, included elsewhere in the Form 8-K. The historical financial information of Inpixon was derived from the unaudited condensed consolidated financial statements of Inpixon and subsidiaries as of and for the nine months ended September 30, 2023 and the audited consolidated financial statements of Inpixon and subsidiaries for the year ended December 31, 2022. Such unaudited pro forma financial information has been prepared on a basis consistent with the financial statements of Legacy XTI and Inpixon and its subsidiaries, respectively. This information should be read together with the financial statements of Inpixon and Legacy XTI related notes, the sections titled "Inpixon Management's Discussion and Analysis of Financial Condition and Results of Operations" and other information included in this Form 8-K, as applicable.

The Merger is anticipated to be accounted for using the acquisition method (as a reverse acquisition), with goodwill and other identifiable intangible assets recorded in accordance with GAAP, as applicable. Under this method of accounting, Inpixon is treated as the "acquired" company for financial reporting purposes. XTI has been determined to be the accounting acquirer because XTI maintains control of the Board of Directors and management of the combined company, and the preexisting shareholders of XTI will have majority voting rights of the combined company. For accounting purposes, the acquirer is the entity that has obtained control of another entity and, thus, consummated a business combination. Under the acquisition method of accounting (as a reverse acquisition), Legacy XTI's assets and liabilities will be recorded at estimated fair value as of the acquisition date. The excess of the purchase price over the estimated fair value of the net assets acquired, if applicable, will be recognized as goodwill. Significant estimates and assumptions were used in determining the preliminary purchase price allocation reflected in the unaudited pro forma condensed combined financial statements. The valuation of the net assets of Inpixon immediately prior to the merger for purposes of presentation within this unaudited pro forma condensed combined financial statements have been prepared based on these preliminary estimates, the final amounts recorded may differ materially from the information presented.

The unaudited pro forma condensed combined balance sheet as of September 30, 2023 combines the historical balance sheets of Legacy XTI and Inpixon on a pro forma basis as if the Merger and related transactions had been consummated on September 30, 2023. The unaudited pro forma condensed combined statements of operations for the nine months ended September 30, 2023 and for the year ended December 31, 2022 give pro forma effect to the Merger and related transactions as if they had occurred on January 1, 2022, the beginning of the earliest period presented. Inpixon and XTI had entered into a promissory note agreement prior to the Merger. Therefore, Inpixon's note receivable and XTI's note payable were eliminated by transaction accounting adjustment F.

These unaudited pro forma condensed combined financial statements are for informational purposes only. They do not purport to indicate the results that would have been obtained had the Merger and related transactions actually been completed on the assumed date or for the periods presented, or which may be realized in the future. The pro forma adjustments are based on the information currently available and the assumptions and estimates underlying the pro forma adjustments are described in the accompanying notes. Actual results may differ materially from the assumptions within the accompanying unaudited pro forma condensed combined financial information.

Description of the Merger Agreement

On July 24, 2023, XTI entered into the Merger Agreement, by and among XTI, Inpixon, and Superfly Merger Sub, Inc., pursuant to which XTI will combine and merge with Merger Sub, whereupon the separate corporate existence of Merger Sub shall cease, and XTI will be the "combined company". On or prior to the Closing Date, Inpixon will effectuate a transaction for the divestiture of its Shoom, SAVES, and Game Your Game ("GYG") lines of business and investment securities, as applicable (the "Solutions Divestiture") by any lawful means, including a sale to one or more third parties, spin-off, plan of arrangement, merger, reorganization, or any combination of the foregoing. Following the Solutions Divestiture, the Inpixon portion of the combined company will be the Industrial Internet of Things ("IIoT") business line.

Pursuant to the Merger Agreement, each share of XTI common stock will be converted into the right to receive a number of shares of Inpixon common stock determined by multiplying such share by the Exchange Ratio determined pursuant to the Exchange Ratio Formula as described in more detail in the section titled "The Merger Agreement - Merger Consideration and Exchange Ratio" included in the registration statement filed on November 7, 2023. Additionally, at the Effective Time, each outstanding option and warrant to purchase shares of XTI common stock will be assumed by the combined company and will be converted into an option or warrant, as applicable, to purchase shares of Inpixon common stock, with necessary adjustments to reflect the Exchange Ratio (collectively, the "Assumed Options and Warrants"). Prior to the Effective Time, all outstanding XTI convertible notes will be converted into XTI common stock and will participate in the merger on the same basis as the other shares of XTI common stock, except for a promissory note in the initial principal amount of \$125,000 which is due within 30 days of closing of the Merger or in January 2024, whichever occurs first.

Each share of Inpixon capital stock, option and warrant to purchase Inpixon common stock that is outstanding at the Effective Time will remain outstanding in accordance with its terms, and such shares of capital stock, options, and warrants will be unaffected by the Merger.

Following the consummation of the Merger, the holders of the outstanding XTI common stock immediately prior to the closing of the Merger will own approximately 77% of the outstanding capital stock of the combined company and the holders of the outstanding capital stock of Inpixon immediately prior to the closing of the Merger will own approximately 23% of the outstanding capital stock of the combined company.

Prior to the consummation of the Merger, Inpixon intends to issue 9,742 shares of a new series of preferred stock at a stated par value of \$1,000, in connection with a holder of the Company's debt converting \$9,741,814 of debt and accrued and unpaid interest into shares of preferred stock. The issuance of the new series of preferred stock is subject to the execution of definitive agreements. See Note O for more details.

Prior to the consummation of the Merger, it is anticipated that XTI will consummate a private placement equity financing of units (each a "Unit" and collectively, the "Units") consisting of (a) one share of XTI common stock and (b) two warrants to purchase one share of XTI common stock each ("XTI Warrants") for aggregate gross proceeds of approximately \$3,800,000 (the "Proposed Offering"). Based on an assumed unit offering price of approximately \$0.34 per unit it is anticipated that 11,299,040 shares of XTI common stock and warrants to purchase 22,458,081 shares of XTI common stock will be issued prior to the consummation of the Merger. The XTI common stock and the XTI Warrants comprising the Proposed Offering will separate upon the Closing and will be issued separately but may only be purchased as a Unit. XTI expects to receive net proceeds of \$3,384,000 for the issuance of the common stock and warrants. The issuance of the XTI common stock and XTI Warrants and final terms of the securities to be issued pursuant to the Proposed Offering are subject to the execution of definitive agreements. See Note P for more details.

Prior to the consummation of the Merger, Inpixon may raise additional funds through the issuance of equity securities. Inpixon has previously executed agreements related to an at-the-market ("ATM") offering and certain warrant financing agreements in place that may be utilized for financing. However, there was no commitment in place at the time these unaudited condensed pro forma financial statements were prepared, and the amount of funds and number of securities issued for additional fundraising is uncertain. Capital raised under these agreements are subject to terms underlying each specific agreement. As such, there is no adjustment included in these pro forma financial statements in relation to the expected future financing events. Inpixon anticipates having a cash and cash equivalents balance of approximately \$2,700,000 when the Merger is consummated. See Note I for more details.

Description of the Solutions Divestiture

On October 23, 2023, a Business Combination Agreement (the "Damon Business Combination Agreement") was entered into by and among Inpixon, Grafiti Holding, Inc., 1444842 B.C. LTD ("Amalco Sub"), and Damon Motors, Inc. ("Damon"), pursuant to which Damon will combine and merge with Amalco Sub, a British Columbia corporation and a wholly-owned subsidiary of Grafiti Holding, Inc., with Damon continuing as the surviving entity and a wholly-owned subsidiary of Grafiti Holding, Inc. (the "Grafiti Holding Transaction").

Pursuant to the Damon Business Combination Agreement, Inpixon formed a newly wholly owned subsidiary, Grafiti Holding, Inc for the sole purpose of consummation of the Grafiti Holding Transaction. Inpixon contributed the assets and liabilities of Inpixon UK, a wholly owned subsidiary of Inpixon, to the then Inpixon wholly owned subsidiary Grafiti Holding, Inc. in accordance with the separation and distribution agreement. As the Registration Statement for the Damon Business Combination Agreement is not expected to become effective until the first half of 2024, on December 27, 2023 Inpixon transferred the Grafiti common shares to a newly-created liquidating trust, titled the Grafiti Holding Inc. Liquidating Trust (the "Trust"), which holds the Grafiti Holding, Inc. common shares for the benefit of the participating Inpixon securityholders. The Grafiti Holding, Inc. common shares will be held by the Trust until the Registration Statement has been declared effective by the Securities and Exchange Commission (the "SEC"). Promptly following the effective time of the Registration Statement, the Trust will deliver the Grafiti Holding, Inc. common shares to the participating Inpixon securityholders, as beneficiaries of the Trust, pro rata in accordance with their ownership of shares or underlying shares of Inpixon common stock as of the record date. Amalco Sub,a wholly-owned, direct subsidiary of Grafiti Holding, Inc., will merge with Damon resulting in Damon as the surviving entity post-merger ("Damon Surviving Corporation"). Upon the consummation of the Merger, both Inpixon UK and Damon will be wholly-owned subsidiaries of Grafiti Holding, Inc., and the ticker symbol will be changed to a symbol to be determined concurrent with the closing.

On February 16, 2024, Inpixon entered into an Equity Purchase Agreement to divest the remaining portion of Shoom, SAVES, and GYG that is excluded from the Grafiti Holding Transaction. The Equity Purchase Agreement, by and among Inpixon ("Seller"), Grafiti LLC, and Nadir Ali ("Buyer"), is structured so that Buyer will purchase from Seller 100% of the equity interest in Grafiti LLC for a minimum purchase price of \$1,000,000 paid in two annual cash installments of \$500,000 due within 60 days after December 31, 2024 and 2025. The purchase price and annual cash installment payments will be (i) increased for 50% of net income after taxes, if any, from the operations of Grafiti LLC for the years ended December 31, 2024 and 2025; (ii) decreased for the amount of transaction expenses assumed; (iii) increased or decreased by the amount working capital of Grafiti LLC on the closing balance sheet is greater or less than \$1,000,000. The Company notes that the estimated purchase price as of the date of this filing was approximately \$1,000,000.

UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET AS OF SEPTEMBER 30, 2023

(in thousands, except share and per share amounts)

ASSETS	INPX (Historical)	Pro Forma Adjustments for Divestiture of Grafiti Holdings Inc. Note 1	Pro Forma Adjustments for Sale of Grafiti LLC Note 2	Pro Forma Adjustments for Subsequent Inpixon Transactions	INPX Pro Forma As Adjusted	XTI (Historical)	Subsequent Financing Transactions of XTI	Transaction Accounting Adjustments	Other Transaction Accounting Adjustments	Autonomous Pro Forma Adjustments Combined
Current assets: Cash and cash equivalents	\$ 13,489	\$ (348)	\$ (811)	\$ 2,356 B (3,000) C	\$ 11,686	\$ 236	\$ 3,384	— J		\$ — \$ 8,678
Accounts receivable, net of allowances Notes and other receivables	1,560 2,210	(46)	(886) (118)	3,000 C	628 6,092	_		(1,723) G — (2,415) F	_	— 628 — 4,027
Inventory Prepaid expenses and other	3,355	_	1,000 (1,353)	- 5,000 C	2,002	_	_	350 F	_	_ 2,002
Current assets Total current assets	1,949 22,563	(396)	(250) (2,418)	2,356	1,697 22,105	13 249	3,384	(8,693)		$\frac{-}{-}$ 1,710 17,045
Property and equipment, net Operating lease right-of-use asset, net	1,013 376	_	(717) (10)	_	296 366	15	_			— 311— 366
Software development costs, net		_	(605)	_	383	_	_	(383) I	_	
Investments in equity securities Long-term investments Intangible assets, net	189 50 2,304	=	(189) (50)	_ 			=			 3,819
Goodwill Other assets Total Assets	164 \$ 27,647	<u></u>	(23)	<u> </u>	141 \$ 25,595		3,384	696 I — \$ (7,124)		- 696 - 141 \$ - \$ 22,378
LIABILITIES AND	\$ 27,047	\$ (396)	\$ (4,012)	\$ 2,356	\$ 25,395	\$ 523	3,384	\$ (7,124)	<u> </u>	<u> </u>
STOCKHOLDERS' EQUITY (DEFICIT)	?									
Current liabilities: Accounts payable Accrued liabilities	\$ 1,920 3,569	\$ (1) (65)	(921) (835)	\$ — (707) A	\$ 998 1,962	\$ 2,372 1,344	_	\$ (1,673) G D — F (608) G (1,150) G (50) G		\$ — \$612 — 1,239
Related party payables Accrued interest	_	_	=	_	=	430 871		— J (259) L (335) N D (502) J	_	— 95 — 270
Customer deposits Convertible and promissory notes - related party, net of unamortized discounts	_	_	_	_	_	1,350	_	(27) F	_	— 1,350
of \$4,495 as of September 30, 2023 Promissory note - 2023						1,768 2,038	72	D (1,079) E (636) J (2,388) F		— 125 — —
Warrant liability Xeriant obligation	1,410	Ξ	=	(491) B	919	460 5,583	Ξ	350 F — (5,583) K	_	_ 1,379
Operating lease obligation, current Deferred revenue	198 1,315	(100)	(7) (682)		191 533	_	_			— 191 — 533
Short-term debt	11,165			(1,861) A	9,304			556 I (1,203) L		
Total current liabilities Operating lease obligation, noncurrent	19,577 188	(166)	(2,445)	(3,059)	13,907 184	16,216	_	(14,587)	(9,742)	_ 5,794 _ 184
SBA loan Convertible and promissory notes - related party, net of unamortized discounts and loan costs of \$1,446,736 as of September 30, 2023	_	_	_	_	_	4,165	_	— (4,165) J	_	_ 65
Total Liabilities	19,765	(166)	(2,449)	(3,059)	14,091	20,446		(18,752)	(9,742)	6,043
Stockholders' Equity (Deficit)										
Series 4 Convertible Preferred Stock Series 5 Convertible	_	_	_	_	_	_	_	_	_	
Preferred Stock Series 9 Nonconvertible Preferred Stock	_	_	_	_	_	_	_	_	9,742 O	
Common Stock	112	_	_	33 A 49 B	194	35	=	39 H 580 I (838) M		_ 16
Additional paid-in capital	358,692	(230)	-	6,159 B 2,535 A	367,156	21,796	3,384	1 E 5 J P 1,078 E 1,561 H (360,361) I 5,298 J 5,583 K 1,462 L	-	— 49,280
Treasury stock Accumulated other	(695) 41	=	=	=	(695) 41	=	=	1,150 G 838 M 335 N 695 I (41) I	[= =

comprehensive (loss)											
income	(2.1=.0=1)		(2.0.00)	(2.251) 7	(0.5.5.4.0.0)			(1.505) 6			(10 =00)
Accumulated deficit	(347,971)	_	(3,860)	(3,361) B	(355,192)	(41,754)	_	(4,297) G	_	_	(42,703)
								(1,600) H			
_								360,140 I			
Stockholders' Equity											
Attributable to Inpixon	10,179	(230)	(3,860)	5,415	11,504	(19,923)	3,384	11,628	9,742	_	16,335
•											
Non-controlling Interest	(2,297)	_	2,297	_	_	_	_	_	_	_	_
Tron controlling interest											
Total stockholders'											
equity (deficit)	7,882	(230)	(1,563)	5,415	11,504	(19,923)	3,384	11,628	9,742	_	16,335
Total Liabilities and	7,002	(250)	(1,505)	3,413	11,504	(17,725)	5,504	11,020	2,742		10,000
Stockholder's Equity			(1045)								
Stockholder's Equity	\$ 27,647 \$	(396) \$	(4,012) \$	2,356	\$ 25,595 \$	523 \$	3,384	\$ (7,124)	<u>s — </u>	<u>s — s</u>	22,378

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE NINE MONTHS ENDED SEPTEMBER 30, 2023

(in thousands, except share and per share amounts)

	INPX (Historical)	Pro Forma Adjustments for Divestiture of Grafiti Holdings Inc. Note 1	Pro Forma Adjustments for Sale of Grafiti LLC Note 2	Pro Forma Adjustments for Subsequent Inpixon Transactions	Spin-Off of CXApp Note 3	INPX Pro Forma As Adjusted	XTI (Historical)	Subsequent Financing Transactions of XTI	Transaction Accounting Adjustments	Other Transaction Accounting Adjustments	Autonomous Entity Adjustments	Pro Forma Combined
Revenues	\$ 7,177	\$ (327)	(3,364)	s —	\$ -	\$ 3,486	s —	_	s —	s —	s —	\$ 3,486
Cost of revenues	1,632	(129)	(385)			1,118						1,118
Gross profit	5,545	(198)	(2,979)			2,368						2,368
Operating expenses:												
Research and												
development	6,380	_	(2,838)	_	_	3,542	1,320	_	_	_	_	4,862
Sales and marketing	3,506	(132)	(1,412)	_	_	1,962	487	_	_	_	_	2,449
General and administrative	13,596	(71)	(1,148)		_	12,377	5,906				(1,305)AA	16,978
Acquisition-related costs	2,343	(/1)	(1,146)			2,343	3,900			_	(785) BB	1,558
Transaction costs	2,970	_	_	_	_	2,970	_	_	_	_	(2,970) BB	3 —
Amortization of	671					(71			(27.6)			205
intangibles Total operating	671					671			(276) DD			395
expenses	29,466	(203)	(5,398)	_	_	23,865	7,713	_	(276)	_	(5,060)	26,242
Loss from operations	(23,921)	5	2,419	_	_	(21,497)		_	276	_	5,060	(23,874)
•												
Other income (expense):												
Interest income	(4.200)		226			(4.064)	(906)		272 1111			(2.216)
(expense), net	(4,300)		236	_		(4,064)	(806)	_	373 HH 1,181 II	_		(3,316)
Interest (expense) -									1,101			
discount accretion	_	_	_	_	_			_	_	_	_	_
Loan cost amortization Change in value of	_	_	_	_	_	_	(66)	_	_	_	_	(66)
Xeriant obligation	_	_	_	_	_	_	(196)	_	196 FF	_	_	_
Change in value of							(170)		.,,			
warrant liability		_	_	_	_		(126)		_	.		(126)
Other (expense) income	1,169	_	_		_	1,169				(1,262) MM		(93)
Unrealized gain/(loss) on equity securities	5,733	_	_	_	_	5,733	_	_	_	(5,733)MM	_	_
Realized loss on equity	5,755					5,755				(0,755)112112		
securities	(6,692)					(6,692)				6,692 MM		
Total other income	(4.000)		226			(2.054)	(1.104)		1.750	(202)		(2.601)
(expense)	(4,090)	_	236	_	_	(3,854)	(1,194)	_	1,750	(303)	_	(3,601)
Net Loss from continuing												
operations, before tax	(28,011)	5	2,655	_	_	(25,351)	(8,907)	_	2,026	(303)	5,060	(27,475)
Income tax provision	(2,488)				2,589	101						101
Net Loss from continuing operations	(30,499)	5	2,655		2,589	(25,250)	(8,907)		2,026	(303)	5,060	(27,374)
Loss from discontinued	(30,499)	3	2,033	_	2,369	(23,230)	(8,907)	_	2,020	(303)	3,000	(27,374)
operations, net of tax	(4,856)	_	0	_	4,856	_	_	_	_	_	_	_
Net Loss	(35,355)	5	2,655	_	7,445	(25,250)	(8,907)	_	2,026	(303)	5,060	(27,374)
Net Loss Attributable to Non-controlling Interest	(1.121)		1,131									
Net Loss Attributable to	(1,131)		1,131									
Stockholders	(34,224)	5	1,524	_	7,445	(25,250)	(8,907)	_	2,026	(303)	5,060	(27,374)
Preferred Dividends										(1,023) LL		(1,023)
Net Loss Attributable to												
Common Stockholders	\$ (34,224)	\$ 5	\$ 1,524	<u> </u>	\$ 7,445	\$ (25,250)	\$ (8,907)	<u>\$</u>	\$ 2,026	\$ (1,326)	\$ 5,060	\$ (28,397)
Net Loss Per Share - Basic												
and Diluted Continuing Operations	\$ (0.82)											\$ (3.35)
Discontinued Operations	\$ (0.82)											\$ (5.55) \$ —
Net Loss Per Share - Basic	(0.14)											
and Dilutive	\$ (0.95)											\$ (3.35)
Weighted Average Shares												
Outstanding												
Basic and Diluted	35,845,916											8,475,135

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE YEAR ENDED DECEMBER 31, 2022

(in thousands, except share and per share amounts)

	INPX (Historical)	Pro Forma Adjustments for Divestiture of Grafiti Holdings Inc.	Pro Forma Adjustments for Sale of Grafiti LLC Note 2	Pro Forma Adjustments for Subsequent Inpixon Transactions	Spin-Off of CXApp Note 3	INPX Pro Forma As Adjusted	XTI (Historical)	Subsequent Financing Transactions of XTI	Transaction Accounting Adjustments	Other Transaction Accounting Adjustments	Autonomous Entity Adjustments	Pro Forma Combined
Revenues	\$ 10,948	\$ (406)		s –	\$ -	\$ 6,109	\$ —	s –	\$ —	s —	s —	\$ 6,109
Cost of revenues	3,425	(40)	(1,265)			2,120						2,120
Gross profit	7,523	(366)	(3,168)	_	_	3,989	_	_	_	_	_	3,989
Operating expenses:												
Research and development	8,338		(3,854)			4,484	2,964					7,448
Sales and marketing	3,876	(203)	(1,459)			2,214	729	_		_	_	2,943
General and		· · ·	, , , ,				10.660				(1.160) 1.1	
administrative Acquisition-related costs	15,520 410	(154)	(1,312)			14,054 410	10,669	_	5,897 1	EE —	(1,169) AA (254) BE	
Impairment of goodwill	6,659	_	(5,476)	_	_	1,183	_	_		_	(1,183) CC	
Amortization of intangibles	1,526	_	(639)	_	_	887	_	_	(362)	on —	_	525
Total operating												
expenses Loss from operations	36,329 (28,806)	(357)			_	23,232 (19,243)	14,362 (14,362)		5,535 (5,535)		(2,606) 2,606	40,523 (36,534)
Loss from operations	(20,000)	(9)	9,372			(19,243)	(14,302)		(3,333)		2,000	(30,334)
Other income (expense): Interest income (expense), net	(677)	_	77	360 G C	· –	(240)	(790)	_	466 I 352		_	(212)
Loan cost amortization	_	_	_	_	_	_	(88)	_		_	_	(88)
Income from stock option forfeitures	_	_	_	_	_	_	14,470	_	_	_	_	14,470
Change in value of												
warrant liability Change in value of	_			_	_	_	12	_	_			12
Xeriant obligation	_	_	_	_	_	_	331	_	(331) 1	FF —	_	_
Other expense, net Unrealized loss on	693	(8)	(63)		_	622	_	_		_		622
equity method												
investment Unrealized gain/(loss)	(1,784)	_	_	_	_	(1,784)		_	_	1,784MM	_	_
on equity securities	(7,904)	_	_	_	_	(7,904)	_		_	7,904MM	_	_
Warrant inducement expense	_	_	_	(3,361) Kk	· _	(3,361)	_	_	_	_	_	(3,361)
Total other income												
(expense)	(9,672)	(8)	14	(3,001)	_	(12,667)	13,935	_	487	9,688		11,443
Net Loss from continuing												
operations, before tax Income tax provision	(38,478) 249	(17) (2)		(3,001)		(38,495) 286	(427)	_	(5,048)	9,688	2,606	(25,091) 286
Net Loss from continuing												
operations Loss from discontinued	(38,229)	(19)	9,625	(3,001)		(38,209)	(427)	_	(5,048)	9,688	2,606	(24,805)
operations, net of tax	(28,075)	_	_	_	28,075	_	_	_	(3,860)	JJ —	_	(3,860)
Net Loss	(66,304)	(19)	9,625	(3,001)	28,075	(38,209)	(427)		(8,908)	9,688	2,606	(28,665)
Net Loss Attributable to Non-controlling Interest	(2,910)	_	2,910	_	_	_	_	_	_	_	_	_
Net Loss Attributable to												
Stockholders Accretion of Series 7	(63,394)	(19)	6,715	(3,001)	28,075	(38,209)	(427)	_	(8,908)	9,688	2,606	(28,665)
Preferred Stock	(4,555)	_	_		_	(4,555)	_		_	_	_	(4,555)
Accretion of Series 8 Preferred Stock	(13,090)	_	_		_	(13,090)	_		_	_	_	(13,090)
Deemed dividend for the	(13,070)					(13,070)						(13,070)
modification related to Series 8 Preferred Stock Deemed contribution for the modification related to Warrants issued in	(2,627)	_	_		_	(2,627)	_		_	_	_	(2,627)
connection with Series 8 Preferred Stock Amortization premium -	1,469	_	_		_	1,469	_		_	_	_	1,469
modification related to	2.625					2.627						2.625
Series 8 Preferred Stock Preferred Stock Dividend	\$ 2,627 \$ —	s –	s —	s —	s –	2,627	_	s –	s —	\$ (974)LL	s —	\$ (974)
Net Loss Attributable to										- (2,1,2)		<u>* (*))</u>
Common Stockholders	\$ (79,570)	\$ (19)	\$ 6,715	\$ (3,001)	\$ 28,075	(54,385)	(427)	<u>s</u> —	\$ (8,908)	\$ 8,714	\$ 2,606	\$ (45,815)
Net Loss Per Share - Basic and Diluted	¢ (22.00)											6 (405)
Continuing Operations Discontinued Operations	\$ (22.08) \$ (12.04)											\$ (4.95) \$ (0.46)
Net Loss Per Share - Basic	(-2.01)											, 5.10
and Dilutive	\$ (34.12)											\$ (5.41)
Weighted Average Shares												
Outstanding												
Basic and Diluted	2,332,041											8,475,135
												

NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

Note 1. Basis of Presentation

The unaudited pro forma condensed combined financial information does not give effect to any anticipated synergies, operating efficiencies, tax savings, or cost savings that may be associated with the Merger. The unaudited pro forma condensed combined financial information is not necessarily indicative of what the actual results of operations and financial position would have been had the Merger and related transactions taken place on the dates indicated, nor are they indicative of the future consolidated results of operations or financial position of the post-combination company. They should be read in conjunction with the historical financial statements and notes thereto of XTI and Inpixon and subsidiaries.

Unaudited Pro Forma Condensed Combined Balance Sheet

- Note 1: Derived from the net assets as of September 30, 2023 of Shoom, SAVES, and Game Your Game that will be divested with Grafiti Holdings Inc. in tandem with the Merger.
- Note 2: Derived from the net assets as of September 30, 2023 of Shoom, SAVES, and Game Your Game that will be divested with Grafiti LLC in tandem with the Merger, along with the inclusion of a \$1,000,000 receivable which represents the estimated sale price of the divestiture.

Unaudited Pro Forma Condensed Combined Statement of Operations

For the Nine Months Ended September 30, 2023

- Note 1: Derived from the statement of operations of Shoom, SAVES, and Game Your Game related to the divestiture of Grafiti Holdings Inc. for the nine months ended September 30, 2023.
- Note 2: Derived from the statement of operations of Shoom, SAVES, and Game Your Game and other related operation activity associated assets related to the sale of Grafiti LLC for the nine months ended September 30, 2023.
- Note 3: To remove discontinued operations related to the CXApp spin-off which was completed in March 2023. Derived from the unaudited condensed consolidated statement of operations of Inpixon and its subsidiaries for the nine months ended September 30, 2023, as presented in the Company's quarterly 10-Q filing. Further adjusted for related impact on deferred taxes as a direct result of the CXApp spin-off.

For the Year Ended December 31, 2022

- Note 1: Derived from the statement of operations of Shoom, SAVES, and Game Your Game related to the divestiture of Grafiti Holdings Inc. for the year ended December 31, 2022.
- Note 2: Derived from the statement of operations of Shoom, SAVES, and Game Your Game and other related operation activity and associated assets related to the sale of Grafiti, LLC for the year ended December 31, 2022.
- Note 3: To remove discontinued operations related to the CXApp spin-off which was completed in March 2023.

Note 2. Accounting Policies and Reclassifications

Upon consummation of the Merger, management will perform a comprehensive review of the two entities' accounting policies. As a result of the review, management may identify differences between the accounting policies of the two entities which, when conformed, could have a material impact on the financial statements of Legacy XTI. Based on its initial analysis, management did not identify any differences that would have a material impact on the unaudited pro forma condensed combined financial information. As a result, the unaudited pro forma condensed combined financial information does not assume any differences in accounting policies.

Note 3. Estimated Purchase Price Consideration

Estimated purchase price of approximately \$20,375,000 related to the Merger is comprised of the following components (in thousands):

Fair value of Common Stock	\$ 9,715
Fair value of Warrants	919
Fair value of Preferred Stock	 9,742
Total Equity Consideration	\$ 20,375

The fair value of common stock of approximately \$9,715,000 included in the total equity consideration is based on Inpixon's closing share price of \$5.00 on February 15, 2024 (which reflects the 1 to 100 reverse stock split which will go effective before the closing of the transaction). The fair value of common stock included in the estimated purchase price will change based on fluctuations in the share price of Inpixon's common stock and the number of equity instruments held by preexisting shareholders of Inpixon on the closing date.

The Merger will be considered a reverse acquisition. As such, the acquisition-date fair value of the consideration transferred is calculated based on the number of equity interests held by Inpixon's preexisting shareholders and retained post-combination. The Company determined the estimated fair value of common stock included in consideration to be calculated based on the Inpixon's common stock outstanding of 1,942,983 multiplied by the price of Inpixon's common stock on February 15, 2024. The Company determined the stock price of Inpixon will be utilized in determining fair value as it is more reliably measurable than the value of the Legacy XTI's (accounting acquirer) equity interests given it is not a publicly traded entity prior to the Merger.

The fair value of warrants of approximately \$919,000 included in the total equity consideration represents 918,689 warrants outstanding by the Company. The fair value of the warrants was determined using level 3 inputs utilizing a Monte-Carlo simulation. The Company determined the fair value of the 918,689 warrants outstanding approximates its redemption value of \$1.00 per warrant.

The fair value of preferred stock of approximately \$9,741,814 included in the total equity consideration represents 9,742 shares of a new series of Preferred Stock that is proposed to be issued and outstanding by the Company upon the consummation of the Merger at a stated value of \$1,000 and fair value of \$1,000 per share. The fair value of the preferred stock was determined using level 3 inputs utilizing a Monte-Carlo simulation.

The fair value of certain consideration related to additional warrants to purchase Inpixon common stock outstanding immediately prior to the consummation of the Merger and equity incentive awards which will remain outstanding with the combined company were not deemed to be significant and were not included in the purchase price consideration for pro forma purposes.

The fair value of common stock included in the estimated purchase price will depend on the market price of Inpixon's common stock when the Merger is consummated. The Company believes that a 10% fluctuation in the market price of its common stock is reasonably possible based on historical volatility, and the potential effect on purchase price would be:

			Fai	r Value of			
			C	Common			
				Stock			
			Included in		Total Equity		
	Comp	ompany's		Consideration		Consideration	
	share price		(in thousands)		(in thousands)		
As presented	\$	5.00	\$	9,715	\$	20,376	
10% increase	\$	5.50	\$	10,686	\$	21,347	
10% decrease	\$	4.50	\$	8,743	\$	19,404	

Note 4. Adjustments to Unaudited Pro Forma Condensed Combined Financial Information

The unaudited pro forma condensed combined financial information has been prepared to illustrate the effect of the Merger and related transactions and has been prepared for informational purposes only.

The following unaudited pro forma condensed combined financial information has been prepared in accordance with Article 11 of Regulation S-X. The unaudited pro forma condensed combined financial information has been prepared to illustrate the effect of the Merger and related transactions and has been prepared for informational purposes only. The Company has elected not to present management adjustments and will only be presenting transaction accounting adjustments and autonomous entity adjustments in the unaudited pro forma condensed combined financial information. The autonomous entity adjustments are management estimates to reflect costs of the IIoT business line being a standalone entity.

The pro forma basic and diluted earnings per share amounts presented in the unaudited pro forma condensed combined statement of operations are based upon the number of shares of the combined company Common Stock outstanding, assuming the Merger and related transactions occurred on January 1, 2022.

Pro Forma Adjustments for Subsequent Inpixon Equity Transactions

The pro forma adjustments for subsequent Inpixon equity transactions represent significant transactions completed by the Company subsequent to September 30, 2023 are as follows:

- A. To account for the issuance of 334,750 shares of common stock issued by Inpixon at various dates between October 1, 2023 to February 14, 2024 related to the conversion of debt and interest of approximately \$2,568,000.
- B. To account for the proceeds for the exercise of warrants with net proceeds of \$2,356,000 inclusive of placement agent fees of \$163,000. The exercise of the liability classified warrants resulted in a decrease in the warrant liability of approximately \$491,000. In connection with the warrant exercise, the Company recorded a warrant inducement expense of \$3,361,000. The warrants were exercised in conjunction with the issuance of new warrants, per the Inducement Agreement entered into on December 15, 2023. The new warrants issued as part of the warrant inducement were determined to equity classified
- C. To account for the purchase of a convertible note and warrants from Damon on October 23, 2023 in an aggregate principal amount of \$3,000,000 for a purchase price of \$3,000,000. The full principal balance and interest on the convertible note will automatically convert into common shares of Damon upon the public listing of Damon.

Pro Forma Adjustments for Subsequent XTI Financing Transactions

The pro forma adjustments for subsequent XTI financing transactions represent significant transactions completed by XTI subsequent to September 30, 2023 are as follows:

D. To reflect the modification of the outstanding Brody 2021 Promissory Note with a new principal balance of \$1,079,000. The difference between the original Brody 2021 Promissory Note principal balance and the new Brody 2021 Promissory Note principal balance is \$72,000, which was the accrued interest of the Brody 2021 Promissory Note.

Transaction Accounting Adjustments to Unaudited Pro Forma Condensed Combined Balance Sheet

The transaction accounting adjustments included in the unaudited pro forma condensed combined balance sheet as of September 30, 2023 are as follows:

- E. To settle XTI's obligation related to Brody's 2021 Promissory Note with a principal balance of \$1,079,000 as of September 30, 2023, which is expected to be settled at the closing of the Merger, provided, however, that up to \$500,000 of the outstanding balance of such note will be converted only to the extent necessary to satisfy applicable Nasdaq initial listing requirements. Assuming full conversion upon closing of the Merger, the principal will be settled through issuance of 1,079,000 common shares of XTI at fair value of \$1,079,000.
- F. To account for the November 13, 2023 amendment of the promissory note entered into between Inpixon and Legacy XTI, which provided for an additional \$700,000 in future loans. On November 14, 2023, Inpixon advanced an additional \$350,000 to Legacy XTI. The adjustment also eliminates the principal balance of the promissory note of \$2,388,000, inclusive of the additional \$350,000 advanced in November 2023, and associated accrued interest of approximately \$27,000. The total amount of \$2,415,000 will be accounted for as an intercompany transaction and eliminated upon consolidation of the combined company. Interest income is reflected net of interest expense on the Unaudited Pro Forma Condensed Combined Statement of Operations, and no pro forma adjustment is required.

- G. Represents estimated non-recurring transaction costs of approximately \$4,297,000 that are expected to be incurred subsequent to September 30, 2023. The estimated transaction costs include advisory, banking, printing, legal and accounting fees, and employee incentive amounts incurred in connection with the Merger. Total estimated transaction costs expected to be incurred by Inpixon and Legacy XTI in connection with the Merger are estimated to be \$4,905,000 and \$4,799,200, respectively. As of September 30, 2023, \$4,407,000 of the total combined estimated costs of \$9,704,200 were accrued by Legacy XTI and Inpixon and were not included in the pro forma adjustment. As such, the pro forma adjustment related to Legacy XTI represents accrued transaction costs of \$949,200. Inpixon estimated transaction costs to be incurred subsequent to September 30, 2023 are expected to be paid in cash at the close of the Merger. Therefore, \$4,905,000 is reflected as a pro forma adjustment to cash. Legacy XTI transaction costs expected to be paid at the close of the Merger in cash total \$1,723,000. Legacy XTI transaction costs expected to be settled at the close of the Merger in equity total \$1,150,000.
- H. Represents the anticipated issuance of common stock to Maxim Group, LLC as payment for its services associated with the Merger, which have a fair value of approximately \$1,600,000.
- I. Represents adjustments for the estimated preliminary purchase price allocation for the Merger. The preliminary calculation of total consideration is presented below as if the Merger was consummated on September 30, 2023:

	Fa	ir Value
	(in t	housands)
Equity consideration ⁽¹⁾	\$	20,375
Total consideration	\$	20,375
Assets acquired:		
Cash and cash equivalents	\$	2,704
Accounts receivable		628
Notes and other receivables		2,092
Inventory		2,003
Prepaid assets and other current assets		1,696
Property and equipment		295
Other assets		507
Tradename & trademarks		722
Proprietary technology		2,409
Customer relationships		470
In-process research and development		348
Goodwill		10,907
Total assets acquired		24,781
Liabilities assumed:		
Accounts payable		995
Accrued liabilities		1,583
Operating lease obligation		376
Deferred revenue		533
Warrant liability		919
Short term debt		<u> </u>
Total liabilities assumed		4,406
Estimated fair value of net assets acquired	\$	20,375

(1) See Note 3

The above purchase price allocation does not give effect to certain pro forma adjustments that were included in the unaudited pro forma condensed combined financial statements that would ultimately impact the purchase price allocation. For any proforma adjustments that were not captured within the closing balance sheet at the time the purchase price allocation was performed, an adjustment was made to goodwill and intangible assets. The impacts of these adjustments decreased goodwill by approximately \$10,211,000 and intangible assets by approximately \$389,000 on the pro forma balance sheet.

Approximately \$696,000 has been allocated to goodwill pursuant to the preliminary purchase price allocation. Goodwill will not be amortized, but instead will be tested for impairment at least annually or more frequently if certain indicators are present. In the event that the value of goodwill or other intangible assets have become impaired, an accounting charge for impairment during the period in which the determination is made may be recognized.

Below is a summary of intangible assets identified and acquired in connection with the Merger based on the preliminary purchase price allocation and the resulting adjustments to recognize the step-up in basis:

		Fair Value	Useful Life
Identified Intangible Assets (in thousands)	Fair Value	Adjustment	(Years)
Tradename & trademarks	\$ 651	\$ 551	5.00
Proprietary technology	2,172	778	7.00
Customer relationships	423	(347)	5.00
In-process research and development	314	314	Indefinite
IP Agreement	_	(40)) N/A
Total	\$ 3,560	\$ 1,256	

An adjustment of \$556,000 was also included to record the assumed debt at fair value, which is due within one year.

This adjustment also eliminates the pro forma historical equity of Inpixon of approximately \$19,327,000 in accordance with the acquisition accounting at closing. This adjustment also reflects the incremental issuance of 6,183,920 shares of the combined company Common Stock, which represents total of 6,532,152 shares of the combined company Common Stock to preexisting XTI shareholders less 348,232 shares outstanding of Legacy XTI that were cancelled and replaced. Additionally, this adjustment accounts for the merger consideration in excess of common stock at closing as additional paid-in capital.

- J. Represents the anticipated conversion of XTI's convertible notes and promissory notes and associated accrued interest to equity in the form of 5,262,000 shares of common stock with a fair value of approximately \$5,303,000 at the date of the Merger. The pro forma adjustment reduces the principal balance and accrued interest balance in the amount of \$4,801,000 and \$502,000, respectively. The only debt expected to remain outstanding is a promissory note that had a principal and accrued interest balance of \$125,000 and \$4,000, respectively, as of September 30, 2023 will also remain outstanding. The promissory note is due within 30 days of closing of the Merger or in January 2024, whichever occurs first.
- K. Represents the conversion of XTI's Xeriant liability to equity of approximately \$5,583,000 on the consummation of the Merger. XTI entered into a joint venture agreement with Xeriant Inc., in which XTI and Xeriant reached an agreement to settle the liability through the issuance of equity.
- L. Represents the conversion of Inpixon's short term debt to equity related to the Solutions Divestiture of approximately \$1,203,000 in principal and \$259,000 of accrued interest, for a total of \$1,462,000, completed during the fourth quarter of 2023.
- M. Represents the impact of Inpixon's 1-to-100 reverse stock split that was approved by Inpixon's board of directors and the shareholders and will go effective before the closing of the transaction.
- N. Represents the relinquishment of an XTI related party payable of \$335,000 that is to be forgiven upon the closing of the transaction.

Other Transaction Accounting Adjustments to Unaudited Pro Forma Condensed Combined Balance Sheet

- O. Represents the issuance of 9,742 shares of a new series of Preferred Stock at stated par value of \$1,000 that is proposed to be issued and outstanding upon the consummation of the Merger. In connection with the closing of the transaction a certain holder of the Company's debt is expected to convert \$9,742,000 of debt and accrued and unpaid interest into shares of Preferred Stock. Based on the analysis of the proposed terms of the preferred stock, the preferred stock is expected to be equity. The final accounting for the issuance of the preferred stock is subject to the execution of definitive agreements and is still under evaluation and may be subject to change.
- P. Represents the issuance of units consisting of 11,299,040 shares of XTI common stock and 22,458,081 XTI warrants in connection with the Proposed Offering. In connection with the closing of the Proposed Offering, the Company is expected to receive funds of \$3,384,000 for the issuance of common stock and warrants, which is net of transaction costs. Based on the analysis of the proposed terms of the warrants, the warrants are expected to be equity. The final accounting for the issuance of the warrants is subject to the execution of definitive agreements and is therefore still under evaluation and may be subject to change.

Autonomous Entity Adjustments to Unaudited Pro Forma Condensed Combined Statements of Operations

The autonomous entity adjustments included in the unaudited pro forma condensed combined statements of operations for the nine months ended September 30, 2023 and the year ended December 31, 2022 are as follows:

- AA. Represents the removal of contractual costs related to Inpixon's CEO, CFO, and internal legal counsel costs for the nine months ended September 30, 2023 and the year ended December 31, 2022 of approximately \$1,305,000 and \$1,169,000, respectively, as these costs will not be incurred by the combined company.
- BB. Represents adjustment to remove \$2,970,000 in expenses that were incurred for the nine months ended September 30, 2023 regarding the CXApp spin-off which was completed in March 2023. In addition, represents adjustment to remove expense related to other non-operational transactions of \$254,000 and \$785,000 for the nine months ended September 30, 2023 and the year ended December 31, 2022, respectively.
- CC. Represents adjustment to remove goodwill impairment that was incurred for the year ended December 31, 2022 of approximately \$1,183,000, as goodwill is reassessed at the date of the transaction. The unaudited proforma statement of operations reflects the transaction as if it had occurred on January 1, 2022.

Transaction Accounting Adjustments to Unaudited Pro Forma Condensed Combined Statements of Operations

The transaction accounting adjustments included in the unaudited pro forma condensed combined statements of operations for the nine months ended September 30, 2023 and the year ended December 31, 2022 are as follows:

DD. Represents incremental adjustments to intangible asset amortization for the step-up in basis of intangible assets subject to amortization acquired in connection with the Merger assuming the Merger occurred on January 1, 2022. The following table is a summary of information related to certain intangible assets acquired, including information used to calculate the amortization expense for each period presented:

Amortization for Period

			Amortization for reflou			100
			Nine	Months		
			Ended		Year	Ended
		Years of	September 30, Decem			nber 31,
Identified Intangible Assets (in thousands)	Fair Value	Amortization	2	023	2	022
Tradename & trademarks	\$ 651	5.00	\$	98	\$	130
Proprietary technology	2,172	7.00		233		310
Customer relationships	423	5.00		63		85
In-process research and development	314	Indefinite		_		_
Total amortization expense			\$	394	\$	525

EE. Reflects the estimated transaction costs of approximately \$11,304,000 to be expensed as if incurred on January 1, 2022, the date the Merger occurred for the purposes of the unaudited pro forma condensed combined statement of operations. This is a non-recurring item. Below represents a summary of the transaction costs associated with the Merger (in thousands).

Third party fees (legal, accounting, investment, etc.)	\$ 2,998
Chardan banker fees - paid in stock	1,000
Bonuses/incentives in connection with Merger	801
Estimated Legacy XTI transaction costs	4,799
Third party fees (legal, accounting, investment, etc.)	2,609
Severance packages in connection with Merger	 2,296
Estimated Inpixon transaction costs	4,905
Maxim Group, LLC banker fees -paid in stock	1,600
Total Estimated Transaction Costs	\$ 11,304

Of the amounts above approximately, \$4,407,000 had previously been expensed. As such, \$6,897,000 was expensed on the statement of operations. See Note G and H for proforma adjustments related to accounting of additional transaction costs expected to be incurred subsequent to September 30, 2023 on the Balance Sheet.

- FF. Represents adjustment to remove the change in fair value related to Legacy XTI's JV obligation which is to be converted into equity at the time of the Merger. The Change in fair value for the nine months ended September 30, 2023 represented a gain of \$196,000. The change in fair value for the year ended December 31, 2022 represented a loss of \$331,000.
- GG. Represents adjustment to record interest income of \$360,000 for the year ended December 31, 2022 regarding the convertible note from Damon outlined in Note D, which has an interest rate of 12% per annum. The note has a term of one year and as such did not include interest income for the nine months ended September 30, 2023 as a pro forma adjustment.
- HH. Represents adjustment to remove interest expense of \$373,000 and \$466,000 related to XTI's convertible notes to be converted at the closing of the Merger, outlined in Note J, for the nine months ended September 30, 2023 and for the year ended December 31, 2022, respectively. The interest expense would not be incurred as a result of the conversion on consummation of the Merger.
- II. Represents adjustments to remove interest expense of \$1,181,000 and \$352,000 related to Inpixon's conversion of debt, outlined in Note A and Note N, for the nine months ended September 30, 2023 and for the year ended December 31, 2022, respectively.
- JJ. Represents adjustment to include the \$3,860,000 loss for the sale of Grafiti LLC outlined in the Description of the Solutions Divestiture section above for the year ended December 31, 2022. Grafiti LLC has net book value of approximately \$2,563,000 as of September 30, 2023, in which the Company had an investment of approximately \$4,860,000. The Company is to receive approximately \$1,000,000 for their ownership interest in Grafiti LLC, resulting in a loss of approximately \$3,860,000.
- KK. Represents adjustment to include warrant inducement expense of \$3,361,000 in connection with the exercise of warrants outlined in Note B.
- LL. Represents adjustments to record preferred dividends and preferred returns of \$1,023,000 and \$974,000 related to the anticipated issuance of a new series of Preferred Stock as outlined in Note N for the nine months ended September 30, 2023 and for the year ended December 31, 2022, respectively. The terms of the new series of Preferred Stock are anticipated to require quarterly dividends beginning on the one year anniversary of the issuance date of the preferred stock, and will be payable on a quarterly basis. The proposed quarterly dividend rate is 1% per quarter and will increase to 3% per quarter on the second year anniversary of the issuance date. The new series of Preferred Stock is also anticipated to require a preferred return on the stated value at the rate of 10% per year, and shall be payable on a quarterly basis.
- MM. Represents adjustments of \$9,688,000 and \$303,000 to remove unrealized and realized gains and losses associated with investments in equity securities that are to be divested with the Grafiti LLC divestiture.

Note 5. Net Loss per Share

Net loss per share was calculated using the historical weighted average shares outstanding, and the issuance of additional shares in connection with the Merger and the related transactions, assuming the shares were outstanding since January 1, 2022. As the Merger and the related transactions are being reflected as if they had occurred at the beginning of the earliest period presented, the calculation of weighted average shares outstanding for basic and diluted net loss per share assumes that the shares issuable relating to the Merger and related transactions have been outstanding for the entirety of all periods presented.

The unaudited pro forma condensed combined financial information has been prepared for the nine months ended September 30, 2023 and for the year ended December 31, 2022 (in thousands, except share and per share data):

	Ser	Nine Months Ended September 30, 2023 (1) Common Stock		September 30, I 2023 ⁽¹⁾		ear Ended cember 31, 2022 ⁽¹⁾ nmon Stock
Pro forma net loss attributable to common stockholders	\$	(28,397)	\$	(45,815)		
Weighted average shares outstanding - basic and diluted		8,475,135		8,475,135		
Pro forma net loss per share attributable to common stockholders - basic and diluted	\$	(3.35)	\$	(5.41)		
Excluded securities: (2)(3)						
Options		846,549		846,549		
Warrants		3,299,383		3,299,383		
Convertible preferred stock		1		1		

- (1) Pro forma net loss per share includes the related pro forma adjustments as referred to within the section "Unaudited Pro Forma Condensed Combined Financial Information."
- (2) The potentially dilutive outstanding securities were excluded from the computation of pro forma net loss per share, basic and diluted, because their effect would have been anti-dilutive. The total amount of dilutive warrants includes 0.9 million unexercised of the 1.5 million warrants that were issued in May 2023.
- (3) The Series 9 Preferred Stock is excluded from the anti-dilutive securities as the holders of the Series 9 Preferred Stock shall not participate in any dividends, distributions, or payments to the holders of Common Stock based on the terms of the securities.